

IAAEU Discussion Paper Series in Economics No. 03/2017

The effect of cigarette taxes during pregnancy on educational outcomes of the next generation

Sonja Settele, Reyn van Ewijk

May 2017

Institute for Labour Law and Industrial Relations in the European Union (IAAEU) 54296 Trier www.iaaeu.de

The effect of cigarette taxes during pregnancy on educational outcomes of the next generation^{*}

Sonja Settele[†] Reyn van Ewijk[‡]

May 18, 2017

Abstract

Smoking during pregnancy is most common among women with a low socioeconomic status and is negatively associated with important infant health measures such as birth weight. Cigarette taxes decrease smoking amongst pregnant women, thereby leading to improved birth outcomes. In this paper we investigate whether increasing cigarette taxes can reduce the intergenerational transmission of a low socioeconomic status by reducing smoking rates among pregnant women with low educational attainment. In a first step, we exploit variation in cigarette taxes across U.S. states over time to show that increasing cigarette taxes leads to improvements in the health of newborns which are larger for babies of low educated mothers. In a second step, we look at subsequent educational success of 16-year-olds measured by grade retention and school enrollment in a large sample of adolescents. We find that increasing cigarette taxes improves the outcomes of children from a low socioeconomic background, but find no effects among children from a higher socioeconomic background. Our findings therefore suggest that cigarette taxes can be an effective policy instrument for mitigating the propagation of a low socioeconomic status from one generation to the next.

JEL Classification: I12, I14, I24.

Keywords: Early Life, Tobacco Taxes, Socioeconomic Inequalities.

^{*}We would like to thank Eric Bettinger, Nadia Campaniello, Diego Cortes Huber, Owen O'Donnell, Christopher Roth, Emilia Soldani, Johannes Wohlfart, seminar participants at the Universities of Frankfurt, Mainz and Karlsruhe as well as participants at the 1st joint seminar by the GSEFM and the IPP and at the IAAEU Workshop on Labor Economics for helpful comments. We also thank Sergej Samoilov for excellent research assistance.

[†]Sonja Settele, Management and Microeconomics Group, Goethe University Frankfurt, e-mail: settele@econ.uni-frankfurt.de

 $^{^{\}ddagger}\mbox{Reyn}$ van Ewijk, Chair of Statistics and Econometrics, Gutenberg University Main, e-mail: vanewijk@unimainz.de

1 Introduction

Important life outcomes such as health, education and income are highly correlated across generations. Consequently, socioeconomic inequalities are persistent (Currie and Hyson, 1999; Currie, 2011; Case et al., 2002; Currie, 2009; Aizer and Currie, 2014). Already by the time of birth, socioeconomic gradients are sizeable. This can be measured by the birth weight of infants, which is the best available proxy for newborn health and a powerful predictor of later life outcomes such as education, later life health (Case et al., 2005; Currie, 2009), life expectancy (Van den Berg et al., 2006; Oreopoulos et al., 2008) and labor market outcomes (Black et al., 2007; Currie and Hyson, 1999; Behrman and Rosenzweig, 2004). Reasons for such early gradients are access to and use of medical care and family planning, environmental factors such as pollution, the mother's health and her nutrition during pregnancy as well as health behaviors such as smoking (Aizer and Currie, 2014; Currie, 2009). Mothers from less advantaged socioeconomic backgrounds on average do worse on all these factors (Cutler and Lleras-Muney, 2010; Phares et al., 2004; Aizer and Currie, 2014).

Public health professionals have identified smoking during pregnancy as the largest risk factor for low birth weight that can be modified by maternal behavior (Kramer, 1987; Shiono and Behrman, 1995). Moreover, socioeconomic differences in this health behavior are strong: Smoking rates in the U.S. are around seven times higher for the lowest income group than for the highest (CDC, 2016a).¹ Smoking is therefore a potential channel for the propagation of socioeconomic inequalities from one generation to the next. Cigarette taxes, in turn, have been shown to reduce smoking during pregnancy, leading to improved birth outcomes (Evans and Ringel, 1999). Reactions to tax increases can be expected to be stronger amongst women with a low socioeconomic status due to a higher incidence of smoking and because poorer women are likely to be more price sensitive than others.

In this paper we study the potential of cigarette taxes in reducing the propagation of socioeconomic inequalities across generations. We do this by analyzing whether cigarette taxes during pregnancy have heterogenous effects across socioeconomic groups at different points in the lifecycle of the offspring. Namely we look at effects on smoking during pregnancy, on infant health and on later life educational outcomes. If cigarette taxes are most effective on the least privileged mothers they will have a diminishing effect on the socioeconomic gradient in smoking behavior. If tax effects translate into improved birth outcomes and later life educational outcomes, then corresponding differences in these outcomes between socioeconomic groups will be reduced as well. One advantage of tobacco taxes as a means of improving relative outcomes of

¹The lowest income group corresponds to less than 10.000 USD income per year (22.3% of mothers smoked during pregnancy), the highest income group is 50.000 USD or more (2.7% of mothers smoked during pregnancy.) These numbers are from 2011.

underprivileged groups is their political implementability. Inequalities can be reduced without a direct redistribution from rich to poor being necessary.

We exploit variation in cigarette excise taxes across U.S. states and over time and use data from two large cross-sectional datasets: The Pregnancy Risk Assessment Monitoring System (PRAMS), providing data on maternal smoking behavior and birth outcomes of live births and the American Community Survey (ACS), providing data on educational outcomes of children and adolescents. In both data sets, information on mothers' educational attainment allows us to analyze heterogenous cigarette tax effects across three broad maternal education groups as proxies for socioeconomic status: i) below high school (less than 12 years education), ii) high school (12 years education), iii) some college or more (more than 12 years education).

We find that cigarette taxes have a strong and significant effect on mothers from the lowest socioeconomic group, i.e. these mothers smoke less in response to increased cigarette taxes. Consequently, their babies benefit in terms of better health at birth. Specifically, we find that a typical tax increase of 10 cents leads to an average weight gain of 2 grams for babies from the lowest socioeconomic group.²

Importantly, there are significant effects at the lower end of the weight distribution where weight gains matter most for infant health and for later life outcomes. For each 10 cents tax increase a baby from our lowest socioeconomic group is 0.07 percentage points less likely to weigh below 2500 grams, the common threshold for a low birth weight. This effect is sizeable, corresponding to more than 10 percent of the difference in the outcome variable between the lowest and the intermediate maternal education group.

In line with the positive cigarette tax effects on birth weight, we find suggestive evidence for corresponding effects on prematurity and on a measure of growth retardation.³ Effect sizes correspond to 5 and 10 percent of the gap to the intermediate socioeconomic group. Especially the latter measure is highly important given that it has been shown to drive the correlation between birth weight and later life health outcomes (Godfrey and Barker, 2001). All effect sizes decrease with maternal education and are economically and statistically insignificant for the highest so-cioeconomic group. We find no evidence that cigarette taxes affect the probability of the baby surviving the first weeks of its life.

Our results also show that improved infant health translates into better educational outcomes

 $^{^2\}mathrm{In}$ our sample period for the PRAMS sample the average cigarette tax increase amounted to 35 cents in real 1998-USD.

 $^{^{3}}$ Growth retardation stands for a low weight relative to the gestational age of the baby.

at age 16. Several proxies for human capital accumulation at age 16 are available all of which are reported by the head of the household, i.e. in most cases the father or mother of the adolescent.

For each tax increase of 10 cents we find a positive effect of around one percentage point on whether a 16-year-old from the lowest socioeconomic group has completed the ninth grade by the time of being surveyed.⁴ Similarly, there are positive effects on a measure of school enrollment and decreasing effects on the probability of grade retention at any time before being surveyed. Lastly, we also provide evidence that the observed tax effects on educational outcomes are at least partially driven by observed cognitive or non-cognitive skills of the adolescent as reported by the head of the household. The effect size of a typical cigarette tax increase of 10 cents corresponds to a reduction in the gap between the lowest and the intermediate socioeconomic group by around 10 percent across all outcomes.

Looking at a restricted set of educational outcomes available for younger children we also show that heterogenous cigarette tax effects are already measurable at age 8 to 15. Taken together, our results suggest that increased cigarette taxes have positive effects on the human capital accumulation of less privileged adolescents, thereby mitigating the transmission of socioeconomic differences to the next generation.

Our results are robust to controlling for a large number of family characteristics as well as state level policies that have been shown to affect infant health and later life educational outcomes. Regarding birth outcomes we show robustness of our results to the quality of prenatal care and other individual-level controls that could systematically differ at the birth cohort level. For the estimated effects on human capital at age 16 we make sure that our results are not spuriously driven by a changing demographic composition of the states' population over time. Moreover, we show robustness to a large range of different specifications. We control for the family's income and mother's and father's employment to make sure that these measures are not driving our results. Furthermore, we control for a large range of state level policies during pregnancy (such as eligibility thresholds for medicaid, smoking bans and the applicable beer tax) and after pregnancy (such as school finance reforms and current cigarette taxes). Even though we allow for the effects of all additional controls to differ by mother's education our estimated cigarette tax effect remains economically and statistically significant.

Previous research gave evidence for a positive effect of cigarette taxes during pregnancy on birth outcomes (Evans and Ringel, 1999; Lien and Evans, 2005) and also showed effects on physical health during childhood and adolescence (Simon, 2016). Our paper is, to the best of our

 $^{^4\}mathrm{In}$ our sample period for the ACS sample the average cigarette tax increase amounted to 12 cents in real 1998-USD.

knowledge, the first to examine the effects of cigarette taxes during pregnancy on the educational success of the next generation and in particular to study their potential in mitigating the propagation of socioeconomic differences from one generation to the next.⁵

In the study most closely related to ours, Simon (2016) uses two cross-sectional datasets in order to analyze the effect of cigarette taxes on birth weight and on the physical health of 2- to 17-year-olds. He finds that a one USD higher cigarette tax during the last term of pregnancy leads to a sizeable decrease in sick days from school and in the probability of having two ore more doctor visits per year. We extend his finding by showing that cigarette taxes also have the potential to improve educational outcomes of the next generation.⁶

At a more general level, our paper also adds to the literature examining the overall effect of cigarette taxes on smoking behavior during pregnancy and on infant health (Evans and Ringel, 1999; Bradford, 2003; Gruber and Köszegi, 2001; Colman et al., 2003; Adams et al., 2012; Lien and Evans, 2005; Philip DeCiccca, 2012).⁷ Some studies have examined heterogenous cigarette tax effects on pregnant women from different socioeconomic groups. They found mixed evidence, with one pointing to higher price elasticities of more highly educated mothers (Ringel and Evans, 2001) and a majority pointing to stronger effects for lower educated mothers (Hawkins and Baum, 2014; Markowitz et al., 2013; Simon, 2016). We complement this literature by looking at a larger number of infant health measures than existing studies. Our results are compelling because all our estimated effects together provide consistent evidence for a positive effect of cigarette taxes on infant health that is stronger amongst lower socioeconomic groups.

We also contribute to a growing literature on the effect of the early life environment on later life educational and other human capital outcomes.⁸ Existing studies exploit variation from natural and policy experiments such as legislation on alcohol availability (Nilsson, forthcoming), the introduction of food stamps (Hoynes et al., 2016), the 1918 influenza pandemic (Almond, 2006), fasting during ramadan (Almond et al., 2015), radioactive pollution due to the 1986 Chernobyl meltdown (Almond et al., 2009) or the Clean Air Act of 1970 (Isen et al., forthcoming). They all point to the importance of the early life environment on measures of human capital across the lifecycle. We study the effect of a specific policy intervention that is easy to influence and

 $^{^5 {\}rm Education}$ is a strong predictor of later life income and other proxies of socioeconomic status (Card, 1999; Waldfogel et al., 2005; Muennig, 2005).

 $^{^{6}}$ The results of Simon (2016) are highly relevant to our study. Especially the tax effect on days of sick leave from school suggests itself as one potential channel through which educational outcomes may be affected by prenatal cigarette tax exposure.

⁷We also contribute to the literature on the causal effects of smoking during pregnancy on infant health that uses identification strategies ranging from controlled experiments (Sexton and Hebel, 1984) to sibling studies (Yan, 2013; Tominey, 2007), minimum ages for cigarette purchase (Yan, 2014) and the 1998 Master Settlement Agreement (Levy and Meara, 2006).

⁸For an overview on the early life origins of human capital development, see also Currie and Almond (2011) and for the early life origins of general life-cycle well-being see Currie and Rossin-Slater (2015).

therefore especially interesting for policy makers.

Another related strand of literature links birth weight, as a proxy for the prenatal environment, to later life educational outcomes. Existing studies use data on twins and siblings (Black et al., 2007; Royer, 2009; Oreopoulos et al., 2008; Currie and Moretti, 2007; Behrman and Rosenzweig, 2004). Effect sizes could however be underestimated due to compensatory prenatal investment or overestimated due to larger perinatal investments into higher birth weight children. Omitted variables could further bias results. Our study adds to this literature by credibly identifying causal effects of cigarette taxes as one important aspect of the early life environment on outcomes at different stages in the life cycle.⁹

Lastly, our paper contributes to analyzing the mechanisms behind the widely observed propagation of socioeconomic status from one generation to the next. The role of the family and socioeconomic background for an individual's human capital development is recognized but not fully understood (Coleman, 1966; Case et al., 2002; Currie and Hyson, 1999). Aizer and Currie (2014) and Currie (2009) propose different environmental and behavioral mechanisms that differ systematically across socioeconomic groups and that are likely to be drivers of the propagation of inequalities acting prior to birth. Our study sheds light on one such mechanism by making use of the arguably exogenous variation in cigarette taxes and analyzing short- and medium-term effects on the next generation.

The rest of the paper is structured as follows: Section 2 provides background information on potential effects of cigarette taxes on the development of the exposed fetus, Section 3 describes the data and section 4 explains the empirical specification. Results are presented in section 5, followed by robustness checks in section 6. Section 7 interprets the magnitude of the estimated effects and Section 8 concludes.

⁹In order to find out what effect prenatal influences have on birth outcomes and how early consequences translate into later life outcomes, a panel data set would be ideal, following individuals exposed to and not exposed to a given influence over the life cycle. Unfortunately, such data are scarce and sample sizes are often too low to detect effects such as those from cigarette taxes. One solution that is possible with existing data is to look at repeated cross sections drawn from the same population, which is our approach. There are few studies to date looking at the effect of the same prenatal influence at more than one point in life. One notable exception is the above-described paper by Simon (2016). Another exception, with data on infant health and later life outcomes drawn from different populations, is the analysis of the effects of in-utero exposure to Ramadan fasting. Almond and Mazumder (2011) find effects of being in utero during Ramadan on birth outcomes using data from the U.S. and on later life health outcomes using data from Uganda and Iraq. Almond et al. (2015) add evidence on later life educational outcomes, using data from England.

2 Background

Cigarette taxes might have a positive effect on educational outcomes via a number of channels. First, smoking during pregnancy could affect the educational outcomes of the offspring by affecting the brain development before birth, and hence directly restricting the offspring's to acquire cognitive and non-cognitive skills during childhood. Second, it may indirectly affect educational outcomes by damaging the baby's physical health prior to birth and thereby impeding cognitive and non-cognitive development later on.

Starting with potential direct effects on the brain development of the fetus, the most relevant factor are the more than 4000 chemicals that are inhaled when smoking a cigarette (Thielen et al., 2008). Some of these chemicals are believed to potentially cause cellular damage through changes in cell structure and hormone levels (Dempsey and Benowitz 2001) and could therefore directly alter the brain development of the fetus. Nicotine, for instance, crosses the placenta and the concentration of nicotine in the embryo can be 15% higher than in the body of the mother (Lambers and Clark, 1996). Nicotine receptors are present very early in the brain of the fetus (Hellström-Lindahl and Nordberg, 2002) and have an important role in brain development (Navarro et al., 1989). Therefore, nicotine could potentially lead to brain damage (Shea and Steiner, 2008). Correlational evidence suggests effects of prenatal tobacco exposure on educational outcomes through delayed or inhibited mental development (Kiechl-Kohlendorfer et al., 2010; Kable et al., 2009) and behavioral outcomes such as physical aggression during early childhood (Huijbregts et al., 2007, 2008) amongst others. Moreover, animal experiments suggest that the cognitive development of animals exposed to nicotine in early life is influenced, even when nicotine levels are too small to have an effect on birth weight (Slotkin, 1998).

Regarding indirect effects, there are several channels through which smoking could have an impact on later life human capital accumulation. First, smoking has an anorexigenic effect, i.e. the mother has less appetite, she might end up eating less and her baby gets less nutrients (Bergen, 2006). Second, smoking leads to a narrowing of blood vessels. Consequently, the blood flow to the placenta is restricted and again, the baby gets less oxygen and nutrients (Lindblad et al., 1988). Third, the inhaled carbon monoxide occupies the hemoglobin in the red blood cells, thereby inhibiting the transportation of oxygen to the fetus (Bureau et al., 1983). All these factors can lead to consequences that will potentially be reflected in birth weight as a proxy for nutritional intake of the fetus. A low birth weight will, on expectation, have consequences over the life-cycle of the offspring (Currie and Hyson, 1999). Lastly, nicotine also hinders the movement of the embryo, which could retard the development of the child's nervous system (CDC, 2010).

3 Data

We use two large cross-sectional data sets covering the different stages of the life cycle we are interested in. First, we use data from the Pregnancy Risk Assessment Monitoring System (PRAMS) on smoking behavior during pregnancy and on birth outcomes reported on a monthly level. Second, we make use of the American Community Survey (ACS) for the analysis of educational outcomes of adolescents. We look at the years 1988-2013 in the PRAMS sample and at 16-yearolds born in 1989-1998 in the ACS.¹⁰¹¹

Our main independent variable are state level cigarette taxes. Nominal cigarette taxes reported on a daily level come from Orzechowski and Walker (2014).¹² We deflate them using the seasonally adjusted monthly CPI provided by the U.S. Bureau of Economic Analysis (2016).¹³

3.1 PRAMS Data

The Pregnancy Risk Assessment Monitoring System (PRAMS), is a comprehensive data set collected by the CDC (2016b) in cooperation with the health departments of participating states. It contains information on birth outcomes and on maternal experiences before, during and shortly after pregnancy for a sample of live births. Mothers are selected based on a stratified sampling design with an oversampling of women at risk for adverse pregnancy outcomes (i.e. low birth weight births, minority races). States are allowed to choose their own stratification criteria but data collection has to follow standardized guidelines developed by the CDC. Probability weights are provided in order to ensure that the sample is representative for the underlying population of recent mothers in the U.S. states that are part of the sample. Sample sizes per state and month generally range between 100 and 250 observations.

The PRAMS data contain information from birth certificates as well information reported by mothers via questionnaires shortly after birth. Data from both sources are combined and only states that reach response rates to questionnaires of at least 70 percent in a given year are included in the PRAMS data set. All measures of infant health we use in our analysis are sourced from birth certificates¹⁴ and are therefore reported with high precision and completeness. The

¹⁰Unfortunately we cannot restrict the PRAMS sample to include the same cohorts as the ACS sample for robustness checks, the reason being that the PRAMS dataset includes too few states in the earlier years. Despite this non-perfect overlap of survey years the PRAMS and the ACS dataset together provide a coherent set of results.

 $^{^{11}\}mathrm{We}$ provide additional evidence by looking at 8-15 year olds and at 17-year olds. The respective cohorts were born in 1990-2006 and 1988-1997.

 $^{^{12}}$ For New York City, which appears as a separate entity in the PRAMS data, we manually added the city tobacco tax to the state-level tax of New York State.

 $^{^{13}}$ The base period was set to June 1998.

¹⁴The only exception is the variable "Baby Alive", which strictly speaking is not a birth outcome variable but a measure for whether the baby was alive at the time of being sampled for the PRAMS dataset.

indicator of smoking behavior is self-reported and will be discussed below.

A total of 38 PRAMS sites granted us access to at least one year of data available during the period 1988 to 2013. We dropped observations where birth weights were below 500 grams because systematic reporting according to the WHO guideline starts at 500 grams. Observations with birth weights above 6000 grams were also dropped, just like twins and other multiple births. We analyze tax effects on the following outcome variables:

- Smoker: A dummy taking on the value one when the mother smoked during pregnancy
- Birth Weight: The birth weight of the newborn measured in grams and grouped in 250g-brackets
- LBW: A dummy for whether the baby's birth weight was below <2500 grams¹⁵
- **Prematurity:** A dummy for prematurity of the baby measured as gestational age below 37 weeks¹⁶
- SGA 10th pctl.: A dummy for whether the baby was small for its gestational age based on the 10th percentile in its gestational age group¹⁷
- **Baby Alive:** A dummy whether the baby was still alive at the time of sampling for the PRAMS dataset.¹⁸

Regarding the smoking dummy we use as our measure of maternal smoking behavior, there are two potential problems. First, there are observations where the smoking dummy is missing (2.4% of the sample). This is not surprising given that such behavior is a sensitive topic. Under our highly restrictive specification, however, there is no significant correlation between the cigarette tax and the probability of the smoking dummy to be missing. From this we can conclude that the missing observations themselves are unlikely to cause bias. Second, non-missing but falsely reported smoking behavior might be an issue. Studies comparing self-reported smoking with results from blood tests report that up to 20-30% of smoking mothers claim they did not smoke during pregnancy (Brachet, 2008; Dietz et al., 2011; Klebanoff et al., 2001). Even though the PRAMS data set was shown to have higher reported smoking rates than other available data sets (Colman et al., 2003), the estimated cigarette tax effect on smoking behavior should be

 $^{^{15}2500}$ grams is the common cutoff below which a newborn is considered to be at risk for severe health and developmental difficulties. (Almond et al., 2005)

 $^{^{16}37}$ weeks is the cut-off for prematurity as defined by the WHO

¹⁷Growth retardation, as measured by a low birth weight relative to gestational age, has not been analyzed in any previous study establishing causality in the context of prenatal smoking exposure/cigarette taxes. It is highly relevant, however, because existing research linking birth weight to later life health outcomes such as coronary heart disease, stroke, hypertension and diabetes has found that associations "depend on lower birth weight in relation to gestation rather than the effects of premature birth" (Godfrey and Barker, 2001)

¹⁸The time at which it is recorded in the dataset whether the baby is alive is a bit unsystematic. The variable can be yes based upon information from the birth certificate (at the time of sampling, limited to 2-6 months after birth, but usually 2-4 months), or when state PRAMS personnel compare the birth certificate number to the infant death registry at the time of mailing (usually within a week or two of the sample).

interpreted with care.¹⁹ We will therefore consider tax effects on birth outcomes as our primary evidence.²⁰

Summary statistics for the outcome variables and the main regressors are given in Table 1. Summary statistics for the large range of other covariates can be found in Table A.1 in the appendix. Figure 2 illustrates the social gradient in smoking during pregnancy and in infant health.

[Insert Table 1 and Figure 2]

3.2 ACS Data

The American Community Survey is conducted once per year by the Census Bureau, providing a representative sample of the U.S. population (Ruggles et al., 2015). The ACS is a repeated cross-sectional dataset covering more than 3 million households and group quarters²¹ per year. Household heads from selected households are asked to provide information on all members of the household. Responses to the ACS are required by law and response rates are exceptionally high²².

We look at the ACS samples for the years 2005 until 2014 because the quarter of birth of each respondent is first reported in 2005. This variable allows us a matching of prenatal cigarette taxes to respondents on a quarterly basis. We first drop all individuals born outside the U.S. to include only those that were exposed to cigarette taxes in a U.S. state. Next, we confine the sample to 16-year-olds born during the first half of the year that live in the same household as their mothers. In Table 2 we provide evidence that our sample of 16-year-olds cannot be distinguished from the full population of 16-year-olds born in the U.S., based on observable characteristics. In Table A.2 of the appendix we show in addition that there is no sample selection based on cigarette taxes.

We include only individuals living with their mother because we want to proxy socioeconomic status by mother's education similar to the PRAMS sample. Maternal characteristics can only be observed for individuals who live in the same household as their mothers. We therefore drop

 $^{^{19}}$ If misreporting is unrelated to cigarette tax increases it should simply add noise to the data and thereby lead to a downward bias of the estimated tax coefficient.

²⁰The PRAMS dataset also includes self-reported information on the number of cigarettes smoked per day. We do not use this information in our regressions due to low reporting quality. Namely, information on the daily quantity of cigarettes is missing for more smokers than non-smokers and in some cases it is entirely censored at the state-level. However, we do use reported information on numbers of cigarettes in order to cross-validate the reported smoking dummy. Specifically, we exclude those observations from the smoking regressions where mothers indicated that they smoked during pregnancy but reported zero cigarettes as well as those where mothers indicating they did not smoke yet reported a strictly positive number of cigarettes per day. These we assume to be reporting errors. They make up less than 0.06 percent of the total sample.

²¹Group quarters include college/university student housing, residential treatment centers, skilled nursing facilities, group homes, military barracks, correctional facilities, workers' group living quarters and Job Corps centers, and emergency and transitional shelters.

 $^{^{22}}$ Usually around 98% of the selected households are included in the final dataset. The year 2013 is an exception: Due to the government shutdown, the ACS did not have a second mailing, a telephone followup, or a person followup for the October 2013 housing unit panel. Therefore, 2013 response rates amount to 90%.

individuals living in group quarters (0.9% of all 16-year olds) and individuals who live in a household but whose mother was not identified in the same household. Educational attainment of the mother is reported for more than 99% of our remaining sample.²³

We focus on 16-year-olds instead of older individuals, first, because school enrollment is obligatory in all states at least until age 16 and has been so throughout the survey years we consider. Consequently, state- and cohort-specific differences in the respective legislation cannot potentially bias our results. Second, at age 16, 88 percent of our remaining sample of 16-year-olds still live in the same household as their mother, which is barely less than at age 10 (90 percent). This share starts to shrink at age 17 and drops to 80 percent by age 18. The decision to move out from home around age 18 may potentially be an outcome affected by cigarette taxes during pregnancy.

We focus at 16-year-olds instead of younger children because in the ACS, educational attainment is only reported in detail starting from grade 9. Also, some of the main educational outcomes reported in the ACS, especially grade retention, get more common as children get older. Nevertheless, in Section 5.2 we also provide evidence on the effect of cigarette taxes on school enrollment and cognitive difficulties among younger age groups.

Lastly, we confine our sample to individuals born in the first and second quarter of the year. Cutoff dates for kindergarten enrollment usually fall between August and December, with a majority in September. For cohorts born in the third and fourth quarter, therefore, changes in these cutoff dates could make a difference in terms of having completed a certain grade at a given age. For children born in the first and second quarter such policy changes should make no difference.²⁴

We look at several outcome variables to measure educational attainment of 16-year-olds:

• Grade 9: A dummy taking on the value one if Grade 9 has been completed by the individual. It takes on the value zero either because the individual is still going to school but has not yet completed grade 9 (i.e. grade retention) or because the individual dropped out of high school before having completed grade 9. Grade 9 completion by the age of 16 is a rather conservative benchmark. However, we believe that a tax effect should be most pronounced at the bottom of the educational achievement distribution.

 $^{^{23}}$ Identified mothers include confirmed mothers as indicated on survey questionnaires (>96 Percent of our sample), but also cases where in the absence of information the census bureau classified female household heads or other adult women living in the same household as most likely mothers (<3 Percent of our sample) and cases where mothers are classified as stepmothers based on marriage to the father (<0.05 Percent). Our results remain unchanged if we include only confirmed mothers.

²⁴It has been shown that relative age within a school class affects academic performance of pupils (Bedard and Dhuey, 2006). Changes in cutoff-dates would slightly change the relative age even of individuals born in January to June. However, there is no reason why cigarette tax increases in utero and changes in cutoff-dates for school enrollment five years later should be systematically correlated across states. Even if there was some spurious correlation, it would be surprising that we find similar effects for different cohorts, looking at 8-11-year-olds, 12-15-year-olds, 16- and 17-year olds.

- Enrolled: A dummy taking on the value one if the individual was enrolled in a regular school²⁵ at any time in the three months before the ACS interview. For individuals that were not enrolled but had already finished Grade 10, the dummy is still set to one. This variable measures the extensive margin for not having completed Grade 9.
- Grade 9|school: A dummy taking on the value one if Grade 9 has been completed by the individual and zero if not but the individual is still enrolled at school. The dummy is set to missing if the individual is not enrolled in any regular school. This variable measures the intensive margin for not having completed Grade 9, i.e. grade retention.
- Cogn. Diff.: A dummy taking on the value one if the individual was reported by the respective household head to have any type of cognitive difficulty. The corresponding questionnaire item reads as follows: "The person has cognitive difficulties (such as learning, remembering, concentrating, or making decisions) because of a physical, mental, or emotional condition."

Summary statistics for our sample of 16-year-olds are shown in Table 3. Figure 3 illustrates the social gradient in educational outcomes at ages 12 and 16. All variables for this figure are coded such that a higher bar stands for a more negative outcome. The bars labelled "retention" stand for the share of 12 (16) year-olds that have not yet completed grade 5 (grade 9) at the time of the survey. Not surprisingly, retention rates increase from age 12 to age 16 because underlying causes are accumulating over time and result in the observed event at some point in time. However, the socioeconomic gap in retention rates is already visible by the age of 12. In contrast, the share of children and adolescents that are reported to not be enrolled in a school is similar across socioeconomic groups at age 12. This suggests that at younger ages, determinants of enrollment are less related to family background. Over the years the socioeconomic gap increases for this outcome as well and by the age of 16, a larger fraction of children from the lowest socioeconomic group have left school than in the other two groups. Cognitive difficulties as perceived by the head of household, however, seem to be visible by age 12 already.

[Insert Table 2, Table 3 and Figure 3]

3.3 State cigarette taxes

In our sample period between 1988 and 2013 there were 206 tax increases by U.S. states, amounting to 35 cents (1998 USD) on average and leading to an increase in real average state level cigarette taxes of more than 700 percent.²⁶ For a detailed description of the tax decision process

 $^{^{25}}$ Regular schools encompass all schools leading toward a high school diploma or college degree. Other types of schooling count only if a regular school or college would have accepted it for credit.

 $^{^{26}}$ The average tax per package increased from 0,19 to 1.63 (1998-USD) over this period. Regarding the shorter sample period of the ACS sample between 1988 and 1998, there were 84 tax increases in the U.S., with an average size of increases of 14 cents (1998 USD).

we refer to Simon (2016) who convincingly argues that cigarette taxes are unlikely to be correlated with sudden shifts in public attitudes towards smoking.²⁷

In order to map the applicable cigarette tax rate to each observation we calculate the date of conception with as much precision as allowed by the data: In the PRAMS-sample, we observe births in different month-year-periods, in the ACS-sample we can estimate quarter-year-periods in which individuals were born. For each individual we compute the average tax rate in the nine months following the estimated date of conception.²⁸

3.3.1 Applicable tax rates for the PRAMS sample

Figure 1 gives an overview of the cigarette tax variation over time for the states (plus New York City) covered by PRAMS data.²⁹

[Insert Figure 1]

In the PRAMS sample, the month and year of birth of the baby are available, as well as a categorical variable on the length of gestation. We assigned each of the five available categories an approximate duration in months, as described in Section B in the appendix and calculate the approximate date of conception, assuming that all babies were conceived in the middle of the respective month. We then used the mean applicable cigarette tax in the state of birth in the nine months after the estimated month of conception.³⁰ Thereby we derive the following cigarette tax rate, applicable for individual i born in state s an conceived at time period t as follows:

$$Tax_{ist} = \frac{1}{9} \sum_{m=t}^{t+8} Tax_{sm}$$

 Tax_{sm} stands for the average tax rate between the 15th of month m and the 15th of month m+1.

 $^{^{27}}$ We refer specifically to the online appendix B of Simon (2016) where the legal processes behind the tax decision process are described and the use of tax funds is outlined.

²⁸Some existing studies exploit tax variation during the last three months of pregnancy because this period was shown to be most important for birth weight effects (United States Department of Health and Human Services, 2001). For the development of cognitive and non-cognitive capacities leading to later life school success, it is unclear, however, which are the sensitive periods during fetal development. The brain and central nervous system of a fetus start to develop already 16 days after conception when the neural plate forms which then evolves into the baby's brain in week six or seven of pregnancy. Several studies suggest that especially smoking early in pregnancy is associated with cognitive damage (Falk et al., 2005; Roza et al., 2007). Similarly, (Almond et al., 2015) show that exposure to Ramadan during the first months of pregnancy has larger negative effects on children's educational performance than exposure during later months.

 $^{^{29}}$ New Hampshire and Iowa joined the PRAMS in 2013 (which is the last year covered by the data) and are therefore not included in the tax graph. Nevertheless, due to the monthly tax variation we exploit, the two states are still included in the respective regressions.

³⁰We decided against using the mean tax during the actual individual months of pregnancy because this could lead to an upward-bias in our estimated tax effect on a range of birth outcomes. For instance a baby that was born prematurely would be assigned a lower cigarette tax and at the same time would have a lower birth weight and gestational age. This correlation would be captured by the tax coefficient.

3.3.2 Applicable tax rates for the ACS sample

The ACS reports the age of each individual at the moment of being surveyed and also their quarter of birth. However, no information on year of birth is given and the exact date of interview is unknown. We know, however, that survey dates are approximately uniformly distributed over the months January to December. This allows us to construct weighted averages of the applicable cigarette tax rates in the two possible years of birth of each individual, with the weights corresponding to the probability of the individual being born in each of the two consecutive years in question.³¹

We know that for seven out of eight individuals born in the first quarter of the year we have BirthYear = SurveyYear - Age because these individuals had their birthday in the survey year before the time of being surveyed. For the remaining one out of eight, the true year of birth is BirthYear = SurveyYear - Age - 1. For those individuals born between April and June, the probability of having one's birthday after the interview date is higher: For them, three out of eight individuals get interviewed before their birthday and five after their birthday. To minimize deviations from true birth dates we assume that all individuals born in Q1 were born on the 15th of February and all individuals born in Q2 were born on the 15th of May (treating all months are equally long). Also, given that we have no information about the length of gestation of each individual, we assume that all individuals were in utero for nine months. Based on these considerations we calculate the applicable tax rate for individual *i* born in state *s* in period *t* as follows:

$$Tax_{ist} = \begin{cases} \frac{7}{8} (\frac{1}{9} \sum_{m=May(y-1)}^{Feb(y)} Tax_{sm}) + \frac{1}{8} (\frac{1}{9} \sum_{m=May(y-2)}^{Feb(y-1)} Tax_{sm}), & \text{if } Q=1 \\ \\ \frac{5}{8} (\frac{1}{9} \sum_{m=Aug(y-1)}^{May(y)} Tax_{sm}) + \frac{3}{8} (\frac{1}{9} \sum_{m=Aug(y-2)}^{May(y-1)} Tax_{sm}), & \text{if } Q=2 \end{cases}$$

where t stands for the time period defined by quarter Q and year y in which the individual was conceived in case the ACS interview took place after the individual's birthday in the survey year. Tax_{sm} stands for the average tax rate between the 15th of month m and the 15th of month m+1. The described way of mapping taxes obviously introduces measurement error to the data. This does not affect the internal validity of our findings, however, given that if anything the estimated cigarette tax coefficient will be biased towards zero.

 $^{^{31}}$ One simply fing assumption we make is that all months have equal length and that birth dates are uniformly distributed over time.

4 Empirical Specification

4.1 General Set-up

We want to analyze how cigarette taxes affect smoking behavior of pregnant women, birth outcomes and educational outcomes of 16-year-olds. Our estimating equation is similar across outcome variables.

For average effects (regardless of socioeconomic background), we estimate the following specification:

$$Y_{ist} = \alpha \operatorname{Tax}_{st} + \Pi^T \mathbf{X}_{ist} + \beta_t + \rho_s + \gamma_s t + u_{ist}$$
(1)

for individual *i*, conceived at time *t* and born in state *s*. Tax_{ist} is the applicable tax during pregnancy, β_t are time of conception dummies, ρ_s are state dummies and $\gamma_s t$ are state-specific time trends. Time units correspond to month-year-periods in the PRAMS-sample and to quarter-yearperiods in the ACS sample. The identifying variation in Tax_{ist} stems from tax changes within states over time that deviate from a linear trend. \mathbf{X}_{ist} is a vector of individual characteristics including mother's age at giving birth, her education at the moment of being surveyed and the individual child's gender. The additional set of controls in \mathbf{X}_{ist} differs depending on the sample. When using the PRAMS data our main specification controls for a large number of dummies covering the following information: Mother's race, her marriage status, a measure of self-reported stressors she experienced during pregnancy, her pregnancy intention³², the Kessner-index for prenatal care as assessed by a doctor³³, the questionnaire phase³⁴. In all regressions using the ACS, respectively, \mathbf{X}_{ist} includes dummies for individual *i*'s race. We do not control for any further characteristics in the ACS-regressions because other available variables were determined after birth and could therefore potentially be affected by cigarette taxes.

Our main interest lies in heterogeneous tax effects across socioeconomic groups as proxied by mothers' educational attainment. For the analysis of these heterogeneous effects we estimate the following specification:

$$Y_{ist} = \alpha_1 \operatorname{Tax}_{ist} + \alpha_2 (\operatorname{mom} \operatorname{HS}) \operatorname{Tax}_{ist} + \alpha_3 (\operatorname{mom} > \operatorname{HS}) \operatorname{Tax}_{ist} + \Pi^T \mathbf{X}_{ist} + \beta_t + \rho_s + \gamma_s t + u_{ist}$$
(2)

where (mom HS) is a dummy taking on the value 1 for all mothers from the intermediate ed-

 $^{^{32}}$ i.e. whether she wanted to become pregnant now, later, sooner, never

³³The Kessner index is reported on the birth certificate on a four-item scale comprising the levels "adequate", "intermediate", "inadequate", "unknown".

 $^{^{34}\}mathrm{The}$ PRAMS questionnaire was reviewed 6 times after its first version in 1988.

ucation group³⁵ and (mom >HS) stands for the highest education group³⁶ respectively. Consequently the coefficient α_1 measures the cigarette tax effect on the lowest education group³⁷. The vector \mathbf{X}_{ist} includes the variables described above. All regressions take account of sampling weights. Standard errors are clustered on state-level.

5 Results

5.1 The effect of cigarette taxes on smoking during pregnancy and birth outcomes

Tables 4 shows the main results for mother's smoking during pregnancy and subsequent birth outcomes, using PRAMS data. Panel A shows that on aggregate, cigarette taxes decrease smoking during pregnancy and thereby improve a range of birth outcomes. For instance the effect in Column (3) suggests that a 10 cent increase in the applicable cigarette tax rate during pregnancy decreases the probability that a baby weighs less than 2500 grams at birth by 0.04 percentage points. Surprisingly, the tax effect on mother's smoking in column (1) is insignificant, while the effect on the probability of having a low birth weight baby in Column (3) is significant. This might be due to noise in the data caused by misreported as described in section 3.1 above.

Table 4 also includes the price elasticities of smoking participation based on our regression results in squared brackets.³⁸ The overall price elasticity of participation in our sample is -0.42. This suggests that a one percent increase in the price of cigarettes reduces the likelihood of an individual woman to smoke during pregnancy by 0.42 percent. This number lies within the range of estimates in the existing literature.

Panel B reports heterogeneous tax effects. The coefficients in the first row represent the effect which cigarette taxes (measured in terms of 10 cent increases) during pregnancy have in the lowest socioeconomic group. The coefficients in the second and third row stand for the difference in the tax effect when the mother has a high school degree and at least some college. Adding up the first and the second coefficient, for instance, yields the cigarette tax effect in the intermediate socioeconomic group.

The estimated coefficients in Columns (1) suggest that mothers from the lowest socioeconomic group are 0.4 percentage points less likely to smoke when the applicable cigarette tax

 $^{^{35}12}$ years of education

 $^{^{36}}$ more than 12 years of education

 $^{^{37}\}mathrm{less}$ than 12 years of education

 $^{^{38}\}mathrm{Details}$ on the calculation of price elasticities can be found in Section C in the appendix.

reate increases by 10 cents. The effect on mothers from the intermediate socioeconomic group are significantly smaller and those for the highest socioeconomic group are close to zero.

The differences are not only driven by a higher share of smokers in the lower socioeconomic groups. Price elasticities reported in squared brackets show that mothers from the lowest socioeconomic group are more than twice as price sensitive as mothers from the intermediate or from the highest socioeconomic group.

Column (2) shows that for each 10 cent increase in cigarette taxes, babies born to the lowest educated mothers gain around two grams on average. Correspondingly, the tax coefficient in Column (3) illustrates that the probability of a baby in the lowest socioeconomic group to weigh less than 2500grams at birth is reduced by 0.07 percentage points. This is quite a substantial effect size, corresponding to more than one tenth of the effect of the mother having a high school degree as compared to being a high school dropout.³⁹

Columns (4) and (5) give further insights into reasons for weight gains. A higher birth weight can be driven by a longer gestation or by a higher birth weight relative to gestational age. Results in Column (4) provide suggestive evidence that babies stay in utero beyond the 37th week of gestation as a consequence of cigarette taxes and thereby have more time to gain important weight whereas Column (5) provides evidence for cigarette taxes mitigating growth retardation in the lowest socioeconomic group.

We find no evidence for a tax effect on whether the baby was still alive at the time of being sampled by the PRAMS, i.e. 2-6 months after birth (Column 6). The coefficients in Column (6) suggest that we do not need to worry about sample selection in our ACS-sample due to selective survival of the strongest babies after birth.⁴⁰

The orders of magnitude of the tax coefficient on all measures of infant health are substantial. Depending on the outcome variable, a tax increase of 10 cents (in real 1998 USD) reduces the socioeconomic health gap between the lowest and the intermediate maternal education group by around 5 to 10 percent.

[Insert Table 4]

 $^{^{39}}$ In the sampling period of the PRAMS dataset the average tax increase was around 35 cents. One in five tax increases even exceeded 50 cents in terms of real-1998-USD. Assuming a linear tax effect, a 35 cents tax increase would reduce the difference between the lowest and the intermediate socioeconomic group in the probability of a low birth weight by more than one third.

⁴⁰Unfortunately we cannot say anything about the effect of cigarette taxes on spontaneous abortions due to the PRAMS sample being restricted to life births. If there was a selection of the strongest infants based on smoking during pregnancy, however, this should lead to a downward bias of our estimated tax coefficient.

5.2 The effect of cigarette taxes on later life educational outcomes

Table 5 displays the results on the effect of prenatal cigarette tax exposure on educational attainment at age 16. All aggregate coefficients in Panel A point into the expected directions, indicating overall higher educational attainment of 16-year-olds due to cigarette taxes and less cognitive difficulties as assessed by the respective head of household. Similar to the infant health outcomes described before, the average effects are insignificant.

Panel B again illustrates heterogeneous tax effects. Column (1), for instance, suggests that for the lowest socioeconomic group a 10 cent tax increase during pregnancy increases the likelihood that the offspring has completed grade 9 by the age of 16 by around one percentage point. This effect size corresponds to around 10 percent of the educational difference to the intermediate socioeconomic group. As expected, estimated tax effects are significantly weaker for the intermediate socioeconomic group and there are no effects on the educational outcomes of the most privileged group.

Figure 4 illustrates this finding, looking at grade 9 completion rates. It is based on a probit regression with the same restrictive specification as the regression results in Table 5. The x-axis spans the 10th to 90th percentile in terms of real cigarette taxes in 1998-USD applicable for our sample of 16-year-olds. The Figure illustrates how the educational achievement gap at age 16 is significantly reduced when cigarette taxes increase.

Results in Columns (2) and (3) of Table 5 shed further light on how exactly cigarette taxes affect the measure of grade 9 completion. Namely, a 10 cents increase in the prenatally experienced cigarette tax leads to a 0.88 percentage points increase in the likelihood of being enrolled in school at the age of 16. Also, it reduces the likelihood of having experienced at least one grade retention prior to being surveyed by 0.65 percentage points. The first of these two effects is very large, especially when considering that school enrollment is obligatory at least until age 16 and that correspondingly, enrollment rates in this age group amount to more than 96 percent, even in the lowest socioeconomic group. The effect size in Column (2) corresponds to a reduction in the difference to the intermedicate socioeconomic group by almost 15 percent, that in Column (3) by a little less than 10 percent.

Finally, Column (4) provides evidence for the role of cognitive difficulties of the individual as reported by the father or mother in shaping the observed effects on educational outcomes. The measure in Column (4) does not distinguish between cognitive aspects such as intelligence and acquired knowledge and non-cognitive aspects such as self-motivation and emotional stability, to name just a few. Nevertheless, it provides credibility to estimated effects in Columns (1) - (3).

[Insert Table 5 and Figure 4]

It is important that the magnitudes of coefficients should not be compared directly between birth outcomes on the one hand (Table 4) and educational outcomes at age 16 on the other hand (Table 5). This is mainly for two reasons:

First, the two measures of socioeconomic status are not directly comparable. In Table 4, mother's educational attainment is measured around the time of giving birth, in Table 5 it is measured 16 years later, at the moment of being surveyed for the ACS. Within these 16 years, a lot of mothers, especially younger ones, complete high school and/or go to college. In the PRAMS sample, around 20 percent of mothers have less than a high school degree, compared to only 10 percent in the ACS. We can therefore expect any cigarette tax effect to be more pronounced in the lowest socioeconomic group of the ACS-sample given that the measure of low socioeconomic status is stricter.

Second, Tables 4 and 5 cover slightly different time periods. Whereas the ACS-sample is largely from the 1990s, the majority of data points in the PRAMS sample come from the years after 2000. One has to consider that average cigarette prices as well as price increases before the year 2000 were relatively low whereas after 2000 a number of large tax hikes was implemented⁴¹.

Figure 5 illustrates the development of cigarette taxes between 1988 and 2013. There is a notable increase in the average cigarette tax as well as in the variation across U.S. states. We can expect that the moderate cigarette tax increases in the 1990s influenced the most price sensitive mothers and that the effect of cigarette taxes was smaller in the 2000s. This argument is supported by empirical findings (Simon, 2016; Levy and Meara, 2006). Nevertheless, we believe that our measured tax effects in both the early life stage and at age 16 give a consistent impression of the potential of cigarette taxes in lowering the propagation of socioeconomic inequalities over generations.

[Insert Figure 5]

We also examine the effect of cigarette taxes on educational and cognitive outcomes at a younger age and for 17-year-olds. Grade completion is not reported on a yearly basis below grade 9. Therefore, in Table 6, we look at cognitive difficulties (Columns (1)-(3)) and school enrollment rates (Columns (4)-(6)) of 8- to 15-year-olds.

 $^{^{41}}$ These differences result in different standard deviations in cigarette taxes across our two samples. Looking at Tables 1 and 3 we see that a standard deviation in cigarette taxes in our earlier ACS sample corresponds to 34 cents whereas in the later PRAMS sample a standard deviation corresponds to 82 cents in real 1998 Dollars.

Whereas the overall tax effect in Panel A is small and insignificant, heterogenous tax effects reported in Panel B show a similar pattern as for the sample of 16-year-olds. Looking at Columns (4) to (6), it is interesting is that both the effect of mother's education and the cigarette tax effect are larger for the slightly older age group of 12- to 15-year olds than for the youngest children. It seems that family background has an increasingly large effect with increasing age and that prenatal smoking is driving some of this effect.

In Table 7 we replicate our main findings for 17-year-olds. The estimated tax effect is similar to that observed among 16-year-olds. If anything, it seems to be a little stronger for the 17-year-olds. The heterogenous tax effect on cognitive difficulties in Column (4) appears insignificant due to large standard errors. The size of the coefficients, however, is comparable to that for 16-year-olds.

[Insert Table 6 and Table 7]

6 Robustness

In this section, we address concerns regarding the causality of our estimated tax effects. There could be omitted variables driving our results, such as policies that vary at the cohort level and influence our outcomes of interest.

6.1 Robustness of the tax effect on smoking and birth outcomes

For the robustness of the early life stage one potential confound could be that revenue generated from cigarette taxes is being used for the financing of the health sector. If this was the case, the observed tax effect would not be caused by changes in prenatal smoke exposure but by an improved quality of state-financed prenatal care, especially for mothers from a relatively low socioeconomic background. For this concern we refer to Simon (2016) who illustrates in detail that most of the cigarette tax revenue in the U.S. goes into the states' general fund or is earmarked for spending unrelated to health care.

We complement his argumentation in Table A.3 where we control for the individually experienced prenatal care of the mother with two different measures. Panel A repeats the baseline specification whereas in Panel B we add a dummy for whether the mother received state-financed supplementary nutrition (WIC) during her pregnancy.⁴² Panel C controls for the Kessner index which is a measure for the quality of prenatal care as assessed by a doctor on the birth certificate.

 $^{^{42}}$ WIC is a program supplementing families at risk of poor pregnancy outcomes with additional healthy nutrition during pregnancy.

We allow all additional controls to vary by mother's education.⁴³

Another potential confound could be that tax hikes might increase the general awareness in the population regarding the dangers of smoking during pregnancy. If this was the case, not the price of cigarettes but the public opinion towards smoking would be driving our results. In Panel D of Table A.3 we show evidence on the robustness of our results to whether the mother remembers being informed by her doctor about the dangers of smoking to her baby.

Our results remain economically and statistically significant in Panels B to D, showing that neither health care financing nor public awareness seem to be a concern for our specification.

Lastly, other tobacco policy changes enacted at the same time as cigarette tax changes could lead to an upward-bias in our estimated tax coefficients. One example for such policy changes are clean indoor air laws that are decided on a state level. Several existing studies have controlled for cigarette taxes and smoking bans at the same time, showing that tax effects on smoking during pregnancy and related birth outcomes are driven by cigarette taxes rather than clean indoor air laws (Simon, 2016; Evans and Ringel, 1999; Gruber and Köszegi, 2001; Markowitz et al., 2013; Hawkins and Baum, 2014; Adams et al., 2012).

[Insert Table A.3]

6.2 Robustness of the tax effect on educational outcomes

6.2.1 Robustness to changes in the demographic composition of the sample

Regarding the estimated tax effect on educational outcomes at age 16 we first make sure that our results are not driven by systematic changes in the sample composition that are correlated with cigarette taxes. Table 8 shows how cigarette taxes are correlated with the demographic composition of our sample across the three categories of maternal education. On average, girls have better school outcomes, just like Whites and Asians.

The tax coefficients in Columns (1) to (3) show that, if anything, the described changes in the sample composition should mitigate the heterogenous tax effect we find. However, most of the tax coefficients are insignificant.

[Insert Table 8]

 $^{^{43}}$ The Kessner Index is already included in our baseline specification. Now we allow its effect to be different according to maternal education group.

6.2.2 Robustness to split samples

In our baseline specification, state-specific time trends do not capture differential trends in education across socioeconomic groups. If, for instance, states that raised cigarette taxes also introduced state policy changes helping less privileged children to succeed in education, these trends could be captured by our tax variable which is allowed to differ by socioeconomic background. In Table 9 we show that our results are robust to splitting our sample by mother's education. In this alternative specification, the state-specific time trends as well as all other control variables are allowed to have different effects across maternal education groups. Still, the cigarette tax effect for the lowest socioeconomic group remains significant whereas for the intermediate and for the highest socioeconomic groups there is no effect.

[Insert Table 9]

6.2.3 Robustness to current economic situation of the family

In Table A.4 we show that our results are robust to the family's current economic situation. A family's economic situation could arguably influence the school success of children. The ACS dataset includes a range of interesting measures at the family level which we again interact with mother's education to allow for differential effects according to the family background.⁴⁴

Panel A shows the baseline results, in Panel B we add the log of the family's real total income, controlling for the number of children and the household size. In Panel C we use a measure of relative income generated by the Census Bureau: It expresses the family's total income as a percentage of the poverty thresholds established by the Social Security Administration. This measure takes into account inflation and depends not only on the total family income but also on the size of the family, the number of people in the family who are children, and the age of the head of household.

In Panels D and E, we add the current employment status of the mother and the father. Our results remain robust, showing that the effect of taxes experienced in utero acts independently of the economic situation of the family. In Panel E, Column (3) effect sizes decrease slightly compared to the baseline results in Panel A, but the general pattern remains the same.

[Insert Table A.4]

 $^{^{44}}$ For instance, an increase in family income for a family with a low socioeconomic status could make a larger difference to the offspring's educational success than the same increase for a family with a high socioeconomic status.

Robustness to other health policies, the business cycle and transfer payments 6.2.4

Table 10 compares our baseline estimates (Panel A) to specifications that control for smoking bans in private worksites, bars and daycare centers (Panel B)⁴⁵, for the relevant beer tax in the in-utero period⁴⁶ (Panel C) and for the current cigarette tax the observed 16-year-olds are exposed to (Panel D) in their state of residence.

In Panel B and C we create averages of the control variables over the 9 months of in-utero period for each individual observation. We then interact each of the newly created variables with the three maternal education groups. Therefore, in Panel B we control for nine and in Panel C for three additional covariates. According to the estimated coefficients in Panel B, our results in Columns (1) and (2) are robust to jointly including all three measures of smoking bans. In Colums (3) and (4) coefficients miss the conventional levels of significance due to large standard errors. Results in Panel C are similar to those in Panel A, suggesting that changes in the beer tax during the in-utero period are not driving our results.

Current cigarette taxes could play a role as confounds because health behavior during youth, specifically the consumption of tobacco, may negatively impact educational attainment via absence from school (Zhao et al., 2012), impaired cognitive development (Jacobsen et al., 2005; Trauth et al., 2000) or physical health (Ding et al., 2009). In Panel D we show that our estimated effect of prenatal cigarette tax exposure is not driven by current cigarette taxes.

[Insert Table 10]

Table A.5 compares our baseline results (Panel A) to the estimated cigarette tax effect when controlling for state-level Medicaid eligibility (Panel B), state welfare reforms (Panel C) and the business cycle (Panel D). Again, we interact all additional controls with maternal education.

States are entitled to determine eligibility criteria for Medicaid (The Kaiser Commission on Medicaid and the Uninsured, 2015). During our sample period, changes in eligibility thresholds were implemented.⁴⁷ These thresholds take the form of income levels relative to the poverty line and there are specific values for pregnant women. We use values provided by Hoynes and Luttmer (2011). Panel A shows that our estimated tax effect remains unaffected by including

⁴⁵Our measures of smoking bans come from ImpacTeen, a research organization on youth health, which rates the stringency of indoor air laws by state on a yearly basis. ⁴⁶Smoking and drinking could be complementary behaviors, i.e. when beer is cheaper people drink more and

consequently also smoke more. Alternatively, one could think of smoking and drinking as substitutes.

⁴⁷Until 1986 families receiving cash assistance were automatically eligible for Medicaid. Beginning in 1987 Medicaid expanded eligibility for children and pregnant women. States were given considerable flexibility to modify their program rules (Mann, 1999) and many states expanded Medicaid beyond the federally mandated levels (Gruber, 2003).

medicaid eligibility thresholds.

During our sample period there were also considerable changes in state welfare systems.⁴⁸ These welfare reforms were found to have negative consequences for the health and health care, especially of unmarried women (Bitler et al., 2005) which could potentially carry over to infant health. We therefore control for a dummy, taking on the value one for completed welfare reform on a state-year level, as defined by Bitler et al. (2005).⁴⁹ Again, our results remain largely unchanged with results in Column (3) being a little more susceptible to the inclusion of additional controls than the rest of the coefficients.

[Insert Table A.5]

Lastly, we also check robustness of our results to controlling for state-level transfer payments in the form of several types of social benefits. Following Simon (2016), we use data from the Regional Economic Information System (REIS), a database which tracks transfer receipts from personal income accounts. We look at per capita expenditure on food stamps⁵⁰, income maintenance⁵¹ and public medical assistance⁵². We construct one measure of average expenditure for each of them for the period of pregnancy up to age five and one measure for age six up to age 16.

In Table A.6 we jointly control for the early childhood and for the later childhood transfer measures. Again, as before, we interact all variables with maternal education, allowing for effects to differ by socioeconomic background. Our main results (Panel A) remain unaffected when controlling for any measure of state-level transfers (Panel B - Panel D).

[Insert Table A.6]

⁴⁸In the period 1993-1996, the Clinton administration granted waivers to 43 states, allowing states to experiment with their welfare systems. Waivers included, among other things, provisions which may require work and/or training, sanctions for those not complying with requirements and limits on the duration of benefit receipt. In addition, in 1996 a law was passed, leading to the replacement of the AFDC program by the TANF program. The primary motivation of all these changes was to decrease the caseload, to help families move quickly from welfare to work, and thus toward self-sufficiency. In the described time period, the welfare caseload dropped dramatically throughout the country (Dion and Pavetti, 2000).

⁴⁹Bitler et al. (2005) define a dummy that takes on the value one whenever a state has either implemented a major waiver, implying a significant deviation from the state's AFDC program or whenever a state has replaced its AFDC program by TANF in a given year. Between 1992 and 1997 all states subsequently implemented welfare reform, according to this definition, with a majority in 1996/97.

 $^{^{50}}$ Exposure to food stamps from the in-utero period up to age five, for instance, were shown to have positive effects on adult health and economic outcomes (Hoynes et al., 2016). Therefore, we control separately for this category, even though it is also included in the category "income maintenance".

 $^{^{51}}$ Income maintenance includes supplemental security income (SSI) benefits, Earned Income Tax Credit (EITC), the Supplemental Nutrition Assistance Program (SNAP), family assistance and other income maintenance benefits

 $^{^{52}}$ By controlling for public medical assistance our aim is to control for influences on the health of children, especially from less privileged backgrounds where access to health care may be severely limited. Therefore we chose to include Medicaid expenditures as well as other medical care benefits but excluded, for instance, Medicare benefits and military medical insurance benefits.

6.2.5 Robustness to changes in the education system

Lastly, we also want to make sure that our results on educational outcomes are robust to changes in the education system. In Panel B of Table 11 we therefore control for state level per pupil expenditures on public elementary and secondary schooling.⁵³ Specifically, we create measures of average expenditures for each individual's school career, i.e. covering the year in which the observed individual reached the school-entry age of five, up to one year before the survey. The estimated cigarette tax effect remains robust even though we allow the education expenditures to have differential effects according to the individual's socioeconomic background.

Moreover, we demonstrate robustness to an exogenous changes in state level education expenditures. Namely, during our sampling period a number of court-mandated changes in school financing principles were exogenously imposed, improving the educational outcomes of children living in poor school districts via better financing (Jackson et al., 2016).⁵⁴ As a consequence of changing state-level financing principles, our cohorts of 16-year-olds were exposed to different education expenditures depending on their state and year of birth.⁵⁵ In Panel D of Table 11 we control for the share of school years each individual spent under the different financing principles. Each of these additional variables is interacted with mother's education, i.e. we include 15 additional variables in this specification.

Panel C in Table 11 shows the baseline results excluding states that are not included in the sample of Jackson et al. (2016).⁵⁶ Panel D shows the actual robustness test. Given the large number of additional control variables it is not surprising that some of the variation in the data is absorbed, resulting in slightly smaller coefficients. The overall picture is, however, that the estimated cigarette tax effect is robust to the described exogenous changes in school financing.

[Insert Table 11]

7 Magnitude of the effect

Our findings indicate that cigarette taxes during the 1990s had a significantly negative effect on socioeconomic difference in human capital accumulation measured at age 16. In this section

 $^{^{53}}$ These data come from the National Public Education Financial Survey (NPEFS), provided by the National Center for Education Statistics.

 $^{^{54}}$ Traditionally the U.S. public education system relied heavily on the local property tax base for school district funding. This started changing in the early 1970s, when a range of major school finance reforms (SFRs) was implemented. Reforms accelerated in the 1980s causing some of the most dramatic changes in the structure of K–12 education spending in U.S. history. Up to the 2000s there were court-mandated financing reforms. Jackson et al. (2016) encoded the court mandated changes into five different principles that are more or less equality-oriented and found significant effects on educational outcomes, especially in poor districts.

 $^{^{55}}$ The assumption that everybody in our sample spends their school years in their state of birth is a simplification. It is true for slightly above 80 percent of our sample of 16-year-olds.

⁵⁶South Dakota and the District of Columbia.

we discuss the magnitude of the estimated effect as well as implications for individuals and society.

Most of the 84 tax increases in the U.S. over the years 1988 to 1998 amounted to only a couple of cents, with an average size of increases of 14 cents. Only three tax hikes in the whole sampling period were of 50 cents or larger. Therefore, for the interpretation of magnitudes we choose to consider the effect of a typical increase of ten cents. Our main regression results suggest that a ten cents tax increase during pregnancy leads to a one percentage point increase in the likelihood of a 16-year-old from the lowest socioeconomic group to have completed grade 9 by the age of 16. This corresponds to roughly ten percent of the difference in this likelihood between 16-year-olds from the lowest socioeconomic group and the intermediate socioeconomic group.

What are the implications and expected long-term benefits for these adolescents? Grade retention is a proxy for underlying skills accumulated over the years. In addition, the event of being retained was shown to have direct negative effects on school success, reinforcing existing disadvantages of low-performing students⁵⁷. We provide some back-of-the-envelope calculations on the personal returns from cigarette taxes per exposed child from the lowest socioeconomic group. (Details of the calculations can be found in the Appendix.)

Each 10 cents tax increase leads to a 1.1 percentage point higher chance of having completed grade 9 by the age of 16. Based on representative statistics, having completed grade 9 by the age of 16 is associated with a roughly 15 percent higher chance of completing high school instead of dropping out (Barro and Kolstad, 1987). We can now calculate the intention-to-treat effect of a 10 cents tax increase on high school completion: The corresponding estimated effect size is 0.165 percentage points for somebody from the lowest socioeconomic group. How does this translate into later life outcomes for the affected person?

Considering expected additional income as well as a lower chance of being unemployed, having a high school degree as compared to having no high school degree is associated with an increase in income of 18,000 USD for men and 14,700 USD for women, again based on representative statistics (Snyder et al., 2016). The corresponding income effect from a 10 cents cigarette tax increase are an annual 30 USD for men and 24 USD for women from the lowest socioeconomic group. These are intention-to-treat effects over the full population of offspring from low educated mothers. The effect on an individual whose mother decided not to smoke based on cigarette taxes

⁵⁷Consider high school completion. In the U.S., the lawful minimum age for leaving school ranges between 17 and 19, depending on the state (Education Commission of the States , ECS). Therefore, a person who had to repeat a grade at some point and then stays in school for as long as legally required will leave school with one completed grade less. A scarce literature establishing causality finds negative effects of grade retention on high school completion in case the retention happens at older ages (Jacob and Lefgren, 2009). Suggested channels are low self-esteem, poor social adjustment, negative attitudes towards school, and problem behaviors (Jimerson et al., 2000; Jimerson and Ferguson, 2007; Nagin et al., 2003).

would be higher.

There are some obvious limitations to these calculations. One we would like to point out is that measures of school success are just proxies for future life success. School success driven by underlying characteristics arguably matters more for life outcomes than educational attainment that was "forced upon" students by external incentives such as schooling laws (Eckstein and Wolpin, 1999). Cigarette taxes during pregnancy exert their effect prior to birth and therefore they must necessarily affect underlying characteristics of children. The numbers we used for our back-of-the-envelope calculations represent plain correlations between schooling and later life outcomes and are therefore likely to understate the positive effects cigarette taxes have on later life outcomes.

Lastly, we want to emphasize that the above calculations consider only personal benefits of the affected cohorts. However, additional benefits for society can be expected from the longterm human capital effects of cigarette taxes. Apart from higher productivity and higher tax payments as a result of higher qualifications, research has shown that higher education is associated with positive attitudes and civic behaviors of individuals (Dee, 2004) including voting and volunteering in their communities (Junn, 2005). Also, higher educated persons are less likely to engage in criminal activity (Raphael, 2006), less likely to rely on public services such as food stamps or housing assistance (Waldfogel et al., 2005), less likely to need government health care (Muennig, 2005) and lastly, also more likely to raise healthier, better-educated children (Wolfe and Haveman, 2002).

8 Conclusion

We investigate whether cigarette taxes are an effective instrument in lowering the propagation of socioeconomic inequalities from one generation to the next. The causal channel we examine runs from maternal smoking behavior during pregnancy via infant health to human capital accumulation by the age of 16, as proxied by school outcomes.

In a first step, we analyze the effect of cigarette taxes experienced during pregnancy on maternal smoking behavior during pregnancy and on birth outcomes. Not surprisingly, cigarette taxes have the strongest effect on the smoking behavior of pregnant women with less than a high school degree. Consequently their babies gain most in terms of infant health. Effect sizes decrease with increasing maternal education. In a second step, we examine whether the effects we observe on birth outcomes are also reflected in educational attainment of young adults in the United States. Indeed, we observe that prenatal exposure to higher cigarette taxes leads to higher educational attainment of those whose mothers have less than a high school degree. These improvements are driven by lower dropout rates and by a lower incidence of grade retention. Moreover, we find that for the lowest socioeconomic group cigarette taxes lead to better outcomes in terms of cognitive and non-cognitive skills assessed by the parents.

Our findings demonstrate that smoking during pregnancy is one channel through which a low socioeconomic status is passed on from one generation to the next. This finding is relevant for policy makers who want to create equal starting conditions for children, regardless of their socioeconomic background. Our results suggest that by lowering smoking rates amongst pregnant women, for instance through increases in cigarette taxes, the persistence of socioeconomic inequalities can be mitigated effectively.

Existing research suggests physical health as one potential channel leading to our observed effects (Simon, 2016). Future research with more detailed data could disentangle different channels behind our estimated tax effect, i.e. cognitive aspects such as intelligence, non-cognitive aspects such as personality traits and physical health. Moreover, it will be interesting to directly study long term cigarette tax effects among adults, for instance by looking at outcomes such as employment, income and adult health.

References

- Adams, E Kathleen, Sara Markowitz, Viji Kannan, Patricia M Dietz, Van T Tong, and Ann M Malarcher, "Reducing prenatal Smoking: the role of state policies," American Journal of Preventive Medicine, 2012, 43, 34–40.
- Aizer, Anna and Janet Currie, "The intergenerational transmission of inequality: Maternal disadvantage and health at birth," *Science*, 2014, 344 (6186), 856–861.
- Almond, Douglas, "Is the 1918 Influenza pandemic over? Long-term effects of in utero Influenza exposure in the post-1940 US population," *Journal of Political Economy*, 2006, 114 (4), 672– 712.
- and Bhashkar A Mazumder, "Health capital and the prenatal environment: the effect of Ramadan observance during pregnancy," *American Economic Journal: Applied Economics*, 2011, 3 (4), 56–85.
- _, Bhashkar Mazumder, and Reyn van Ewijk, "In utero Ramadan exposure and children's academic performance," *The Economic Journal*, 2015, *125* (589), 1501–1533.
- _, Kenneth Y Chay, and David S Lee, "The Costs of Low Birth Weight," The Quarterly Journal of Economics, 2005, 120 (3), 1031–1083.
- _ , Lena Edlund, and Mårten Palme, "Chernobyl's subclinical legacy: prenatal exposure to radioactive fallout and school outcomes in sweden," *Quarterly Journal of Economics*, 2009, 124 (4), 1729–1772.
- Barro, Stephen M and Andrew J Kolstad, Who drops out of high school?: Findings from high school and beyond, Center for Education Statistics, Office of Educational Research and Improvement, US Department of Education, 1987.
- Bedard, Kelly and Elizabeth Dhuey, "The persistence of early childhood maturity: International evidence of long-run age effects," *Quarterly Journal of Economics*, 2006, pp. 1437–1472.
- Behrman, Jere R and Mark R Rosenzweig, "Returns to birthweight," Review of Economics and Statistics, 2004, 86 (2), 586–601.
- Bergen, Hugo T, "Exposure to smoke during development: fetal programming of adult disease," Tobacco Induced Diseases, 2006, 3 (2), 1.
- Bitler, Marianne P, Jonah B Gelbach, and Hilary W Hoynes, "Welfare reform and health," *Journal of Human Resources*, 2005, 40 (2), 309–334.
- Black, Sandra E, Paul J Devereux, and Kjell G Salvanes, "From the cradle to the labor market? the effect of birth weight on adult outcomes," *Quarterly Journal of Economics*, 2007, 122 (1).

Brachet, Tanguy, "Maternal Smoking, Misclassification, and Infant Health," 2008.

- Bradford, W David, "Pregnancy and the Demand for Cigarettes," American Economic Review, 2003, 93 (5), 1752–1763.
- Bureau, MA, D Shapcott, Y Berthiaume, J Monette, D Blouin, P Blanchard, R Begin, and N Gagnon, "Maternal cigarette smoking and fetal oxygen transport: a study of P50, 2, 3-diphosphoglycerate, total hemoglobin, hematocrit, and type F hemoglobin in fetal blood," *Pediatrics*, 1983, 72 (1), 22–26.
- **Card, David**, "The causal effect of education on earnings," *Handbook of Labor Economics*, 1999, *3*, 1801–1863.
- Case, Anne, Angela Fertig, and Christina Paxson, "The lasting impact of childhood health and circumstance," *Journal of Health Economics*, 2005, 24 (2), 365–389.
- -, Darren Lubotsky, and Christina Paxson, "Economic status and health in childhood: The origins of the gradient," American Economic Review, 2002, 92 (5), 1308–1334.
- **CDC**, "How tobacco smoke causes disease: the biology and behavioral basis for smokingattributable disease: a report of the surgeon general," 2010.
- _, "Dataset: Maternal Behavior/Health, Tobacco use, indicator for whether mother smoked during the last three months of pregnancy, data from 2011," freely available data set 2016.
- _, "Dataset: Pregnancy Risk Assessment Monitoring System (PRAMS), data from 2016, covering the years 1988-2013," 2016.
- Coleman, James, "Equity of Education Opportunity," Washington D.C.: U.S. Government Printing Office, 1966.
- Colman, Greg, Michael Grossman, and Ted Joyce, "The effect of cigarette excise taxes on smoking before, during and after pregnancy," *Journal of Health Economics*, 2003, 22 (6), 1053–1072.
- Currie, Janet, "Healthy, wealthy, and wise: Socioeconomic status, poor health in childhood, and human capital development," *Journal of Economic Literature*, 2009, 47 (1), 87–122.
- _, "Inequality at Birth: Some Causes and Consequences," American Economic Review, 2011, 101 (3), 1–22.
- and Douglas Almond, "Human capital development before age five," Handbook of Labor Economics, 2011, 4, 1315–1486.
- and Enrico Moretti, "Biology as Destiny? Short-and Long-Run Determinants of Intergenerational Transmission of Birth Weight," *Journal of Labor Economics*, 2007, 25 (2), 231–264.

- and Maya Rossin-Slater, "Early-life origins of life-cycle well-being: Research and policy implications," *Journal of Policy Analysis and Management*, 2015, 34 (1), 208–242.
- and Rosemary Hyson, "Is the Impact of Health Shocks Cushioned by Socioeconomic Status? The Case of Low Birthweight," *American Economic Review*, 1999, 89 (2), 245–250.
- Cutler, David M and Adriana Lleras-Muney, "Understanding differences in health behaviors by education," *Journal of Health Economics*, 2010, 29 (1), 1–28.
- **DeCiccca, Justin Smith Philip**, "Can higher cigarette taxes still improve birth outcomes? Evidence from recent large increases," *LCERPA Working Paper*, 2012.
- **Dee, Thomas S**, "Are there civic returns to education?," *Journal of Public Economics*, 2004, 88 (9), 1697–1720.
- den Berg, Gerard J Van, Maarten Lindeboom, and France Portrait, "Economic conditions early in life and individual mortality," *American Economic Review*, 2006, 96 (1), 290–302.
- Dietz, Patricia M, David Homa, Lucinda J England, Kim Burley, Van T Tong, Shanta R Dube, and John T Bernert, "Estimates of nondisclosure of cigarette smoking among pregnant and nonpregnant women of reproductive age in the United States," American Journal of Epidemiology, 2011, 173 (3), 355–359.
- Ding, Weili, Steven F Lehrer, J Niels Rosenquist, and Janet Audrain-McGovern, "The impact of poor health on academic performance: New evidence using genetic markers," *Journal of Health Economics*, 2009, 28 (3), 578–597.
- Dion, M Robin and LaDonna Pavetti, "Access to and Participation in Medicaid and the Food Stamp Program: A Review of the Recent Literature," Technical Report, Mathematica Policy Research 2000.
- Eckstein, Zvi and Kenneth I Wolpin, "Why youths drop out of high school: The impact of preferences, opportunities, and abilities," *Econometrica*, 1999, 67 (6), 1295–1339.
- Education Commission of the States (ECS), "Compulsory School Age Requirements," 2010.
- Evans, William N and Jeanne S Ringel, "Can higher cigarette taxes improve birth outcomes?," *Journal of Public Economics*, 1999, 72 (1), 135–154.
- _ , _ , and Diana Stech, "Tobacco taxes and public policy to discourage smoking," in "Tax Policy and the Economy, Volume 13," MIT Press, 1999, pp. 1–56.
- Falk, L, A Nordberg, Å Seiger, A Kjaeldgaard, and E Hellström-Lindahl, "Smoking during early pregnancy affects the expression pattern of both nicotinic and muscarinic acetylcholine receptors in human first trimester brainstem and cerebellum," *Neuroscience*, 2005, 132 (2), 389–397.

- Federal Trade Commission, "Competition and the financial impact of the proposed tobacco industry settlement," 1997.
- Godfrey, Keith M and David JP Barker, "Fetal programming and adult health," Public Health Nutrition, 2001, 4 (2b), 611–624.
- **Gruber, Jonathan**, "Medicaid," in "Means-tested transfer programs in the United States," University of Chicago Press, 2003, pp. 15–78.
- and Botond Köszegi, "Is Addiction "Rational"? Theory and Evidence," Quarterly Journal of Economics, 2001, 116 (4), 1261–1303.
- Hawkins, Summer Sherburne and Christopher F Baum, "Impact of state cigarette taxes on disparities in maternal smoking during pregnancy," *American Journal of Public Health*, 2014, 104 (8), 1464–1470.
- Hellström-Lindahl, Ewa and Agneta Nordberg, "Smoking during pregnancy: a way to transfer the addiction to the next generation?," *Respiration*, 2002, 69 (4), 289–293.
- Hoynes, Hilary, Diane Whitmore Schanzenbach, and Douglas Almond, "Long-run impacts of childhood access to the safety net," *American Economic Review*, 2016, 106 (4), 903–934.
- Hoynes, Hilary W and Erzo FP Luttmer, "The insurance value of state tax-and-transfer programs," *Journal of Public Economics*, 2011, 95 (11), 1466–1484.
- Huijbregts, Stephan CJ, Jean R Séguin, Mark Zoccolillo, Michel Boivin, and Richard E Tremblay, "Associations of maternal prenatal smoking with early childhood physical aggression, hyperactivity-impulsivity, and their co-occurrence," *Journal of Abnormal Child Psychology*, 2007, 35 (2), 203–215.
- _ , _ , _ , _ , _ , and _ , "Maternal prenatal smoking, parental antisocial behavior, and early childhood physical aggression," *Development and Psychopathology*, 2008, 20 (02), 437–453.
- Isen, Adam, Maya Rossin-Slater, and W Reed Walker, "Forthcoming." Every Breath You Take—Every Dollar You'll Make: The Long-Term Consequences of the Clean Air Act of 1970."," Journal of Political Economy, forthcoming.
- Jackson, C Kirabo, Rucker C Johnson, and Claudia Persico, "The Effects of School Spending on Educational and Economic Outcomes: Evidence from School Finance Reforms," *Quarterly Journal of Economics*, 2016, 131 (1), 157–218.
- Jacob, Brian A and Lars Lefgren, "The effect of grade retention on high school completion," American Economic Journal: Applied Economics, 2009, pp. 33–58.

- Jacobsen, Leslie K, John H Krystal, W Einar Mencl, Michael Westerveld, Stephen J Frost, and Kenneth R Pugh, "Effects of smoking and smoking abstinence on cognition in adolescent tobacco smokers," *Biological Psychiatry*, 2005, 57 (1), 56–66.
- Jimerson, Shane, Byron Egeland, L Alan Sroufe, and Betty Carlson, "A prospective longitudinal study of high school dropouts examining multiple predictors across development," *Journal of School Psychology*, 2000, 38 (6), 525–549.
- Jimerson, Shane R and Phillip Ferguson, "A longitudinal study of grade retention: Academic and behavioral outcomes of retained students through adolescence.," School Psychology Quarterly, 2007, 22 (3), 314.
- Junn, Jane, "The political costs of unequal education," in "Campaign for Educational Equity Fall 2005 Symposium on the "Social Costs of an Inadequate Education," Teachers College, Columbia University" 2005.
- Kable, Julie A, Claire D Coles, Mary Ellen Lynch, and Julie Carroll, "The impact of maternal smoking on fast auditory brainstem responses," *Neurotoxicology and Teratology*, 2009, 31 (4), 216–224.
- Keeler, Theodore E, Teh wei Hu, Paul G Barnett, Willard G Manning, and Hai-Yen Sung, "Do cigarette producers price-discriminate by state? An empirical analysis of local cigarette pricing and taxation," *Journal of Health Economics*, 1996, 15 (4), 499–512.
- Kiechl-Kohlendorfer, U, E Ralser, Ursula Pupp Peglow, G Reiter, E Griesmaier, and R Trawöger, "Smoking in pregnancy: a risk factor for adverse neurodevelopmental outcome in preterm infants?," Acta Paediatrica, 2010, 99 (7), 1016–1019.
- Klebanoff, Mark A, Richard J Levine, Cynthia D Morris, John C Hauth, Baha M Sibai, Ben Curetevine, Patrick Catalano, and Diana G Wilkins, "Accuracy of selfreported cigarette smoking among pregnant women in the 1990s," *Paediatric and perinatal epidemiology*, 2001, 15 (2), 140–143.
- Kramer, Michael S, "Determinants of low birth weight: methodological assessment and metaanalysis.," Bulletin of the World Health Organization, 1987, 65 (5), 663.
- Lambers, Donna S and Kenneth E Clark, "The maternal and fetal physiologic effects of nicotine," in "Seminars in Perinatology," Vol. 20 Elsevier 1996, pp. 115–126.
- Levy, Douglas E and Ellen Meara, "The effect of the 1998 Master Settlement Agreement on prenatal smoking," *Journal of Health Economics*, 2006, 25 (2), 276–294.
- Lien, Diana S and William N Evans, "Estimating the impact of large cigarette tax hikes the case of maternal smoking and infant birth weight," *Journal of Human Resources*, 2005, 40 (2), 373–392.

- Lindblad, Annika, Karel Maršál, and Karl-Erik Andersson, "Effect of nicotine on human fetal blood flow.," Obstetrics & Gynecology, 1988, 72 (3), 371–382.
- Mann, Cindy, The ins and outs of delinking: Promoting Medicaid enrollment of children who are moving in and out of the TANF System, Center on Budget and Policy Priorities, 1999.
- Markowitz, Sara, E Kathleen Adams, Patricia M Dietz, Viji Kannan, Van T Tong et al., "Tobacco Control Policies, Birth Outcomes, and Maternal Human Capital," *Journal of Human Capital*, 2013, 7 (2), 130–160.
- Muennig, Peter, "Health returns to education interventions," in "symposium on "The Social Costs of Inadequate Education" (Columbia University Teachers College" 2005.
- Nagin, Daniel S, Linda Pagani, Richard E Tremblay, and Frank Vitaro, "Life course turning points: The effect of grade retention on physical aggression," *Development and Psychopathology*, 2003, 15 (02), 343–361.
- Navarro, HA, FJ Seidler, JP Eylers, FE Baker, SS Dobbins, SE Lappi, and TA Slotkin, "Effects of prenatal nicotine exposure on development of central and peripheral cholinergic neurotransmitter systems. Evidence for cholinergic trophic influences in developing brain.," Journal of Pharmacology and Experimental Therapeutics, 1989, 251 (3), 894–900.
- Nilsson, Peter, "Alcohol Availability, Prenatal Conditions, and Long-term Economic Outcomes," *Journal of Political Economy*, forthcoming.
- Oreopoulos, Philip, Mark Stabile, Randy Walld, and Leslie L Roos, "Short-, medium-, and long-term consequences of poor infant health an analysis using siblings and twins," *Journal* of Human Resources, 2008, 43 (1), 88–138.
- Orzechowski, W. and R. Walker, *The Tax Burden on Tobacco, Historical Compilation, vol.* 49, Orzechowski and Walker, Economic Consulting Firm, 2014.
- Phares, Tanya M, Brian Morrow, Amy Lansky, Wanda D Barfield, Cheryl B Prince, Kristen S Marchi, Paula A Braveman, Letitia M Williams, and Brooke Kinniburgh, "Surveillance for disparities in maternal health-related behaviors—selected states, Pregnancy Risk Assessment Monitoring System (PRAMS), 2000–2001," MMWR (Morbidity and Mortality Weekly Report) Surveillance Summary, 2004, 53 (4), 1–13.
- Pregnancy Risk Assessment Monitoring System(PRAMS), data set 2016.
- Raphael, Steven, "The socioeconomic status of black males: The increasing importance of incarceration," *Public Policy and the Income Distribution*, 2006, pp. 319–358.
- Ringel, Jeanne S and William N Evans, "Cigarette taxes and smoking during pregnancy," American Journal of Public Health, 2001, 91 (11), 1851–1856.

- Royer, Heather, "Separated at girth: US twin estimates of the effects of birth weight," American Economic Journal: Applied Economics, 2009, 1 (1), 49–85.
- Roza, Sabine J, Bero O Verburg, Vincent WV Jaddoe, Albert Hofman, Johan P Mackenbach, Eric AP Steegers, Jacqueline Witteman, Frank C Verhulst, and Henning Tiemeier, "Effects of maternal smoking in pregnancy on prenatal brain development. The Generation R Study," *European Journal of Neuroscience*, 2007, 25 (3), 611–617.
- Ruggles, Steven, Katie Genadek, Ronald Goeken, Josiah Grover, and Matthew Sobek, Integrated Public Use Microdata Series: Version 6.0 [dataset], Minneapolis: University of Minnesota, http://doi.org/10.18128/D010.V6.0 2015.
- Sexton, Mary and J Richard Hebel, "A clinical trial of change in maternal smoking and its effect on birth weight," *Jama*, 1984, *251* (7), 911–915.
- Shea, Alison K and Meir Steiner, "Cigarette smoking during pregnancy," Nicotine & Tobacco Research, 2008, 10 (2), 267–278.
- Shiono, Patricia H and Richard E Behrman, "Low birth weight: Analysis and recommendations," The Future of Children, 1995, pp. 4–18.
- Simon, David, "Does early life exposure to cigarette smoke permanently harm childhood welfare? Evidence from cigarette tax hikes," American Economic Journal: Applied Economics, 2016.
- Slotkin, Theodore A, "Fetal nicotine or cocaine exposure: which one is worse?," Journal of Pharmacology and Experimental Therapeutics, 1998, 285 (3), 931–945.
- Snyder, Thomas D, Cristobal de Brey, and Sally A Dillow, "Digest of Education Statistics 2014, NCES 2016-006.," National Center for Education Statistics, 2016.
- Sullivan, Daniel, "Testing hypotheses about firm behavior in the cigarette industry," Journal of Political Economy, 1985, 93 (3), 586–598.
- Sumner, Daniel A, "Measurement of monopoly behavior: an application to the cigarette industry," *Journal of Political Economy*, 1981, pp. 1010–1019.
- The Kaiser Commission on Medicaid and the Uninsured, "Medicaid Financing: How does it work and what are the implications?," online article May 2015.
- Thielen, Anja, Hubert Klus, and Lutz Müller, "Tobacco smoke: unraveling a controversial subject," *Experimental and Toxicologic Pathology*, 2008, 60 (2), 141–156.
- **Tominey, Emma**, *Maternal smoking during pregnancy and early child outcomes*, Centre for Economic Performance, London School of Economics and Political Science, 2007.

- Trauth, Jennifer A, Everett C McCook, Frederic J Seidler, and Theodore A Slotkin, "Modeling adolescent nicotine exposure: effects on cholinergic systems in rat brain regions," *Brain Research*, 2000, 873 (1), 18–25.
- **United States Department of Health and Human Services**, "Women and Smoking: A Report of the Surgeon General," *Centers for Disease Control and Prevention. Atlanta*, 2001.
- U.S. Bureau of Economic Analysis, "Consumer Price Index of All Urban Consumers: All Items (CPIAUCSL)," freely available data set June 2016. Data accessed in June 2016. The Consumer Price Index for All Urban Consumers: All Items (CPIAUCSL) is a measure of the average monthly change in the price for goods and services paid by urban consumers between any two time periods.
- Waldfogel, Jane, Irwin Garfinkel, and Brendan Kelly, "Public Assistance Programs: How Much Could Be Saved with Improved Education?," in "symposium on "The Social Costs of Inadequate Education" (Columbia University Teachers College" 2005.
- Wolfe, Barbara and Robert Haveman, "Social and Nonmarket Benefits from Education in an Advanced Economy," in "Conference Series; Proceedings," Vol. 47 Federal Reserve Bank of Boston 2002, pp. 97–142.
- Yan, Ji, "Prenatal Smoking Cessation and Infant Health: Evidence from Sibling Births," Southern Economic Journal, 2013, 80 (2), 299–323.
- _ , "The Effects of a Minimum Cigarette Purchase Age of 21 on Prenatal Smoking and Infant Health," *Eastern Economic Journal*, 2014, 40 (3), 289–308.
- Zhao, Meng, Yoshifumi Konishi, and Paul Glewwe, "Does Smoking Affect Schooling? Evidence from Teenagers in Rural China.," *Journal of Health Economics*, 2012, 31 (4), 584– 598.

Main Tables

Variable	Mean	Std. Dev.	Min.	Max.	Ν
Cig. Tax	0.82	0.72	0.04	4.32	683828
Smoker	11.62	32.05	0	100	668578
Birth Weight	3332.86	567.44	625	5875	679882
LBW $(<2500g)$	6.17	24.06	0	100	679882
Prematurity $(<37 \text{ wks.})$	7.96	27.07	0	100	675604
Rel. LBW (<10 th pctl.)	9.90	29.88	0	100	662030
Baby Alive	99.74	5.12	0	100	614491
0-11 yrs educ (< HS)	18.22	38.60	0	100	673128
12 yrs educ (HS)	29.94	45.80	0	100	673128
13+ yrs educ (>HS)	51.84	49.97	0	100	673128

Table 1: Summary statistics for PRAMS sample (1988-2013): main variables

Means and standard deviations of cigarette taxes are reported in USD. Birth weight is reported in grams. For all other variables, means and standard deviations are reported in percent of the sample.

	Age 16						
	F	Full ACS Sample			Subsample Living with Mother		
Variable	Mean	Std. Dev.	Ν	Mean	Std. Dev.	Ν	
Grade 9	94.38	23.02	391,731	95.02	21.75	338,012	
Enrolled	98.80	10.91	391,731	98.98	10.04	338,012	
Grade 9 school	94.83	22.14	$381,\!192$	95.41	20.93	$330,\!503$	
Cogn. Diff.	4.694	21.15	391,731	4.246	20.16	338,012	
Born Q1 (Jan-Mar)	24.34	42.91	391,731	24.33	42.91	338,012	
Born Q2 (Apr-Jun)	24.97	43.29	391,731	25.05	43.33	338,012	
Born Q3 (Jul-Sep)	25.99	43.86	391,731	25.98	43.85	338,012	
Born Q4 (Oct-Dec)	24.70	43.13	391,731	24.64	43.09	338,012	
Female	48.76	49.98	391,731	49.29	50.00	338,012	
Race: Amer. Indian/ Alaska Native	2.069	14.23	391,731	1.890	13.62	338,012	
Race: Asian	3.908	19.38	391,731	4.070	19.76	338,012	
Race: Black or African American	17.19	37.73	391,731	16.00	36.66	338,012	
Race: Pacific Islander	0.386	6.201	391,731	0.377	6.131	338,012	
Race: White	74.80	43.41	391,731	76.06	42.67	338,012	
Race: Other Race	5.945	23.65	391,731	5.854	23.48	338,012	

Table 2: Summary statistics for full ACS and our subsample (means and st. dev. in percent)

Table 3: Summary statistics for ACS sample (1988-1998): main variables

Variable	Mean	Std. Dev.	Min.	Max.	Ν
Cig. Tax	0.339	0.156	0.0253	0.860	391,731
$\mathrm{mom} > \mathrm{HS}$	55.13	49.74	0	100	$335,\!239$
$\mathrm{mom} < \mathrm{HS}$	9.883	29.84	0	100	$335,\!239$
mom HS	34.99	47.69	0	100	$335,\!239$
Grade 9	94.38	23.02	0	100	391,731
Enrolled	98.80	10.91	0	100	391,731
Grade 9 school	94.83	22.14	0	100	381,192
Cogn. Diff.	4.694	21.15	0	100	391,731

Means and standard deviations of cigarette taxes are reported in USD. For all other variables, means and standard deviations are reported in percent of the sample. The numbers in this table refer to the full sample of 16-year-olds born between 88/89 and 97/98.

Table 1. Tobacco tax enect on showing during pregnancy and on birth outcomes							
	(1)	(2)	(3)	(4)	(5)	(6)	
VARIABLES	Smoker	Birth Weight	LBW	Prematurity	SGA 10th pctl.	Baby Alive	
Panel A: Aggregate Tax Effect							
Cig. Tax	-0.119	0.503	-0.043***	-0.014	-0.021	-0.006	
	(0.086)	(0.501)	(0.012)	(0.030)	(0.027)	(0.012)	
mom HS	-5.154^{***}	29.306***	-0.441***	-0.303***	-1.126***	0.037^{*}	
	(1.019)	(4.284)	(0.115)	(0.093)	(0.257)	(0.019)	
$\mathrm{mom} > \mathrm{HS}$	-13.145***	63.274***	-1.454***	-0.973***	-2.678***	0.128***	
	(1.486)	(5.284)	(0.133)	(0.126)	(0.337)	(0.022)	
Mean of dep. var.	11.52	3335	6.103	7.901	9.811	99.75	
Panel B: Heterogeneous Tax Effect	;						
Cig. Tax	-0.39***	1.96***	-0.07***	-0.05	-0.08*	-0.00	
Price elasticity (mom <hs)< td=""><td>(0.08) [-0.76]</td><td>(0.61)</td><td>(0.02)</td><td>(0.03)</td><td>(0.04)</td><td>(0.01)</td></hs)<>	(0.08) [-0.76]	(0.61)	(0.02)	(0.03)	(0.04)	(0.01)	
Tax x (mom HS)	0.27***	-0.80*	0.02	0.04*	0.02	0.00	
	(0.06)	(0.45)	(0.02)	(0.02)	(0.03)	(0.00)	
Price elasticity (mom HS)	[-0.29]						
Tax x (mom $>$ HS)	0.36***	-2.18***	0.04**	0.04*	0.09**	-0.00	
· · · · ·	(0.12)	(0.47)	(0.02)	(0.02)	(0.04)	(0.00)	
Price elasticity (mom $>$ HS)	[-0.23]	· · · ·	· · /			· · /	
mom HS	-7.23***	35.63***	-0.60***	-0.60***	-1.31***	0.03	
	(1.18)	(5.83)	(0.16)	(0.19)	(0.38)	(0.02)	
mom > HS	-16.05***	81.00***	-1.78***	-1.31***	-3.41***	0.15***	
	(1.81)	(6.76)	(0.17)	(0.19)	(0.47)	(0.04)	
Mean outc. <hs< td=""><td>20.93</td><td>3236</td><td>8.118</td><td>9.438</td><td>12.89</td><td>99.64</td></hs<>	20.93	3236	8.118	9.438	12.89	99.64	
Mean outc. HS	16.73	3302	6.964	8.521	10.88	99.70	
Mean outc. >HS	5.391	3386	4.932	7.031	8.165	99.82	
Observations	615,794	624.387	624,387	620,828	621,519	561,563	
State FE	\checkmark	\checkmark	1	\checkmark	1	√ √	
Time FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	
State spec. Cohort Trends	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	
Baseline Controls	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	

Table 4: Tobacco	tax effect c	on smoking	during pregnancy	and on	birth outcomes
			0 F - 0 /		

The eigarette tax variable corresponds to the mean of the eigarette tax rate in the 9 months following the month of conception. It is deflated to real mid-1998 USD using a monthly consumer price index and scaled such that coefficients correspond to the effect of 10-cent eigarette tax increases. Columns 1 and 3-6 look at different dummy outcome variables. Dummies are coded as 0-100 for better readability so that coefficients can be interpreted as percentage points. Column 1 in addition includes price elasticities of smoking participation per maternal education group in squared brackets. All specifications control for state fixed effects, time fixed effects as well as state-specific time trends (state dummy*time). All specifications control for a large set of controls: baby's gender, mother's age (7 age groups), pregnancy intention (5 groups), number of stressors experienced during pregnancy (4 groups), questionnaire phases (7 phases). Standard errors are clustered by state*year. Panel A shows aggregate results. Panel B shows results with interaction terms between applicable eigarette taxes and dummies for mother's education group. Coefficients for all other variables are still aggregate. Robust standard errors clustered on state-level are in parenthesis. * p < 0.10, ** p < 0.05, *** p < 0.01.

	(1)	(2)	(3)	(4)
VARIABLES	Grade 9	enrolled	Grade9 school	Cogn. Diff.
Panel A: Aggregate Tax Effect				
Cig. Tax	0.11	0.04	0.13	-0.10
mom HS	(0.13) 7.54***	(0.06) 2.78^{***}	(0.12) 6.10^{***}	(0.14) -1.66***
mom >HS	(1.28) 9.92^{***}	(0.95) 3.42^{***}	(0.87) 8.25^{***}	(0.61) -2.66***
Mean outc.	(1.38) 95.16	(1.00) 99.07	$(0.99) \\ 95.51$	$(0.70) \\ 4.219$
Panel B: Heterogeneous Tax Effect				
Cig. Tax	1.10^{**}	0.88^{***}	0.65^{*}	-0.58*
Tax x (mom HS)	(0.44) -0.98*	(0.32) -0.89**	(0.35) -0.46 (0.27)	(0.33) 0.51^{**}
Tax x (mom >HS)	(0.50) -1.18**	(0.35) -0.96***	(0.37) -0.66*	(0.25) 0.57^{**}
mom HS	(0.49) 10.98^{***}	(0.35) 5.91^{***}	(0.38) 7.74^{***}	(0.27) -3.43***
mom >HS	(2.13) 14.04^{***}	(1.72) 6.77^{***}	(1.30) 10.55^{***}	(0.85) -4.64***
	(2.05)	(1.74)	(1.34)	(0.85)
Mean outc. <hs< td=""><td>86.55</td><td>96.37</td><td>88.12</td><td>6.067</td></hs<>	86.55	96.37	88.12	6.067
Mean outc. HS	94.27 07.21	98.93 00.62	94.58	4.748
Mean outc. >HS	97.21	99.62	97.32	3.308
Observations	$165,\!260$	165,260	$161,\!697$	165,260
Cohorts	89-98	89-98	89-98	89-98
State FE	\checkmark	\checkmark	\checkmark	\checkmark
Time FE	\checkmark	\checkmark	\checkmark	\checkmark
State spec. Cohort Trends	√	~	V	\checkmark
Baseline Controls	\checkmark	\checkmark	\checkmark	\checkmark

Table 5: Tobacco tax effect on educational outcomes at	Table 5:	Tobacco tax	effect on	educational	outcomes at	age 16
--	----------	-------------	-----------	-------------	-------------	--------

The cigarette tax variable corresponds to the mean of the cigarette tax rate in the 9 months following the estimated date of conception. It is deflated to real mid-1998 USD using a monthly consumer price index and scaled such that coefficients correspond to the effect of 10-cent cigarette tax increases. All outcome variables are dummies, coded as 0-100 for better readability so that coefficients can be interpreted as percentage points. All specifications control for state fixed effects, time fixed effects, state-specific time trends, gender, race and mother's age at giving birth. Panel A shows aggregate results. Panel B shows results with interaction terms between applicable cigarette taxes and dummies for mother's education group. Coefficients for all other variables are still aggregate. Robust standard errors clustered on state-level are in parenthesis. * p < 0.10, ** p < 0.05, *** p < 0.01.

			(-)		(-)	(-)
	(1)	(2)	(3)	(4)	(5)	(6)
		Cogn. Diff.			Enrolled	
Age Group	8-15	8-11	12-15	8-15	8-11	12-15
Panel A: Aggregate Tax Effect						
Cig. Tax	-0.03	-0.04	-0.10	-0.01	0.01 (0.02)	0.00 (0.03)
mom HS	(0.46)	(0.01) -1.03** (0.44)	(0.50) -1.34** (0.51)	(0.02) 0.33 (0.22)	(0.02) -0.01 (0.11)	(0.69^{*}) (0.36)
mom >HS	(0.10) -2.30^{***} (0.57)	(0.12) -2.30^{***} (0.59)	-2.29^{***} (0.57)	(0.22) (0.22)	(0.12) (0.12)	(0.36) (0.36)
Mean outc.	4.624	4.696	4.552	98.06	98.26	97.85
Panel B: Heterogeneous Tax Effect						
Cig. Tax	-0.19^{*}	-0.18^{**}	-0.32	0.09^{*}	0.03	0.21^{**}
Tax x (mom HS)	0.15 (0.10)	0.11 (0.10)	(0.20) (0.21) (0.14)	-0.10^{**} (0.05)	-0.02 (0.03)	-0.25^{**} (0.11)
Tax x (mom >HS)	0.19 (0.12)	0.17 (0.11)	0.27 (0.17)	-0.11^{**} (0.05)	-0.03 (0.02)	-0.24^{**} (0.11)
mom HS	-1.83^{***} (0.50)	-1.57^{**} (0.61)	-2.16^{***} (0.54)	0.77^{**} (0.35)	0.07 (0.18)	1.63^{**} (0.64)
mom >HS	-3.12^{***} (0.54)	-3.10^{***} (0.64)	-3.30^{***} (0.58)	1.18^{***} (0.34)	0.46^{***} (0.17)	2.01^{***} (0.64)
Mean outc. <hs< td=""><td>6.159</td><td>6.220</td><td>6.097</td><td>98.06</td><td>98.26</td><td>97.85</td></hs<>	6.159	6.220	6.097	98.06	98.26	97.85
Mean outc. HS Mean outc. >HS	$5.346 \\ 3.940$	$5.562 \\ 3.950$	$5.142 \\ 3.931$	$98.30 \\ 98.69$	98.17 98.52	$98.42 \\98.86$
Cohorts	1990-2006	1994-2006	1990-2002	1990-2006	1994-2006	1990-2002
State FE Time FE	√ √	√ √	√ √	√ √	√ √	√ √
State spec. Cohort Trends	v V	v V	v V	v V	v V	v V
Baseline Controls	√	√	√	√	√	√
Age FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark

Table 6: Tobacco tax effect on educational outcomes of 8- to 15-year-olds

	(1)	(2)	(3)	(4)
VARIABLES	Grade 10	enrolled	Grade10 school	Cogn. Diff.
Panel A: Aggregate Tax Effect				
Cig. Tax	0.25^{*}	0.16^{**}	0.11	0.00
mom HS	(0.13) 9.34^{***} (1.82)	(0.08) 3.66^{***} (1.06)	(0.13) 6.89^{***} (1.27)	(0.12) -1.36* (0.68)
$\mathrm{mom} > \mathrm{HS}$	(1.62) 12.00^{***} (1.08)	(1.00) 4.48^{***} (1.10)	(1.27) 8.92^{***} (1.41)	(0.08) -2.30^{***} (0.70)
Mean outc.	(1.98) 94.50	(1.10) 98.90	(1.41) 95.41	(0.79) 4.221
Panel B: Heterogeneous Tax Effect				
Cig. Tax	1.80^{***}	1.09^{***}	1.10^{**}	-0.47
Tax x (mom HS)	(0.52) -1.60***	(0.33) -0.95*** (0.22)	(0.42) -1.04**	(0.30) 0.64^{*}
Tax x (mom >HS)	(0.55) -1.79***	(0.33) -1.08*** (0.22)	(0.44) -1.12** (0.55)	(0.34) 0.46 (0.26)
mom HS	(0.64) 14.70^{***}	(0.33) 6.85^{***}	(0.55) 10.44^{***}	(0.36) -3.49** (1.20)
mom > HS	(2.36) 17.99^{***} (2.64)	(1.71) 8.11*** (1.70)	$(1.78) \\ 12.73^{***} \\ (2.06)$	(1.36) -3.85*** (1.35)
Mean outc. <hs Mean outc. HS Mean outc. >HS</hs 	84.01 93.59 96.85	95.26 98.72 99.62	87.32 94.59 97.16	$5.745 \\ 4.710 \\ 3.652$
Cohorts State FE Time FE State spec. Cohort Trends Baseline Controls	88-97 ✓ ✓ ✓	88-97 ✓ ✓ ✓	88-97 ✓ ✓ ✓	88-97 ✓ ✓
Observations	161,781	161,781	155,541	161,781

Table 7: Tobacco tax effect on educational outcomes at age 17

The cigarette tax variable corresponds to the mean of the cigarette tax rate in the 9 months following the estimated date of conception. It is deflated to real mid-1998 USD using a monthly consumer price index and scaled such that coefficients correspond to the effect of 10-cent cigarette tax increases. All outcome variables are dummies, coded as 0-100 for better readability so that coefficients can be interpreted as percentage points. All specifications control for state fixed effects, time fixed effects as well as state-specific time trends. In addition, all specifications control for gender, race and mother's age at giving birth. Panel A shows aggregate results. Panel B shows results with interaction terms between the applicable cigarette tax and dummies for mother's education group. Coefficients for all other variables are still aggregate. Robust standard errors clustered on state-level are reported in parenthesis. * p < 0.10, ** p < 0.05, *** p < 0.01.

	(1)	(\mathbf{a})		
	(1)	(2)	(3)	(4)
	Rel. family	Female	White	Asian
VARIABLES	income			
Cig. Tax	-2.59*	-0.23	-0.62	-0.62
	(1.53)	(0.35)	(1.05)	(0.40)
$Tax x \pmod{HS}$	2.43	0.12	1.00	0.37
	(1.48)	(0.43)	(0.92)	(0.29)
$Tax x \pmod{>HS}$	4.40^{***}	0.15	0.97	0.73^{*}
	(1.50)	(0.37)	(1.06)	(0.40)
mom HS	79.96***	0.15	6.61^{***}	-0.49
	(5.19)	(1.72)	(2.29)	(0.72)
$\mathrm{mom} > \mathrm{HS}$	153.26***	-0.01	9.95***	-0.25
	(5.47)	(1.42)	(2.64)	(0.94)
Observations	165.260	165.260	165.260	165.260
State FE	\checkmark	\checkmark	\checkmark	\checkmark
Year of birth FE	\checkmark	\checkmark	\checkmark	\checkmark
Quarter of Birth FE	1	\checkmark	\checkmark	\checkmark
Assessment Year FE	1	1	1	1
State spec. Cohort Trends	√	√	✓	√
Gender	<u> </u>	X	1	1
Bace			x	x
Mother's Age at giving birth	· ·	, ,	,. ,	, ,

Table 8: Demographic composition of the sample of 16-year-olds

The outcome variables in columns (2) to (5) are dummy variables, the outcome variable in column (1) is a continuous variable expressing total family income in percent above the poverty threshold. All regressions are based on our standard specification, controlling for state fixed effects, time fixed effects, state-specific time trends (state dummy*time), mother's education and her age at giving birth. Standard errors are clustered on state level.* p < 0.10, ** p < 0.05, *** p < 0.01.

	(1)	(2)	(3)	(4)
VARIABLES	Grade 9	enrolled	Grade9 school	Cogn. Diff.
Panel A: Baseline Results				
Cig. Tax	1.10**	0.88***	0.65*	-0.58*
Tax x (mom HS)	(0.44) - 0.98^*	(0.32) - 0.89^{**}	(0.35) -0.46	(0.33) 0.51^{**}
Tor r (mom > HS)	(0.50)	(0.35)	(0.37)	(0.25)
1ax x (110111 > 115)	(0.49)	(0.35)	(0.38)	(0.27)
Observations	165,260	165,260	161,697	165,260
Panel B: Split Sample <hs< td=""><td></td><td></td><td></td><td></td></hs<>				
Cig. Tax	1.31^{**} (0.57)	0.66^{*} (0.38)	1.27^{**} (0.54)	-0.46 (0.80)
Observations	13,962	13,962	13,066	13,962
Panel C: Split Sample HS				
Cig. Tax	-0.13 (0.22)	-0.05 (0.10)	-0.09 (0.20)	$0.04 \\ (0.17)$
Observations	$57,\!480$	$57,\!480$	56,086	$57,\!480$
Panel D: Split Sample >HS				
Cig. Tax	$0.02 \\ (0.12)$	-0.01 (0.04)	$\begin{array}{c} 0.06 \\ (0.12) \end{array}$	-0.10 (0.12)
Observations	93,818	93,818	92,545	93,818
State FE Time FE	√ √	√ √	√ .<	√ √
State spec. Cohort Trends	v V	v V	v v	v v
Baseline Controls	√	√	\checkmark	√

Table 9: Split samples: Cigarette tax effect on educational outcomes at age 16

The cigarette tax variable corresponds to the mean of the cigarette tax rate in the 9 months following the estimated date of conception. It is deflated to real mid-1998 USD using a monthly consumer price index and scaled such that coefficients correspond to the effect of 10-cent cigarette tax increases. All outcome variables are dummies, coded as 0-100 for better readability so that coefficients can be interpreted as percentage points. All specifications control for state fixed effects, time fixed effects as well as state-specific time trends. In addition, all specifications control for gender, race and mother's age at giving birth. Panel A shows the baseline results. Panels B, C and D show results for separate regressions by maternal education group. Robust standard errors clustered on state-level are in parenthesis. * p < 0.10, ** p < 0.05, *** p < 0.01.

	(1)	(2)	(3)	(4)
VARIABLES	Grade 9	enrolled	Grade9 school	Cogn. Diff.
Panel A: Baseline Results				
Cig. Tax	1.10**	0.88***	0.65*	-0.58*
Tax x (mom HS)	(0.44) -0.98*	(0.32) -0.89**	(0.35) -0.46 (0.27)	(0.33) 0.51^{**}
Tax x (mom >HS)	(0.50) -1.18** (0.40)	(0.35) - 0.96^{***}	(0.37) -0.66* (0.28)	(0.25) 0.57^{**} (0.27)
Observations	(0.49) 165,260	(0.55) 165,260	(0.38) 161,697	(0.27) 165,260
Panel B: Robustness to smoking bans				
Cig. Tax	1.07**	0.87***	0.60	-0.58
Tax x (mom HS)	(0.44) -0.97*	(0.31) -0.93***	(0.36) -0.42 (0.27)	(0.36) 0.58^{**}
Tax x (mom >HS)	(0.49) -1.10** (0.47)	(0.34) -0.97*** (0.34)	(0.37) -0.55 (0.38)	(0.27) 0.60^{**} (0.29)
Bans in private worksites x Mother's Educ	\checkmark	\checkmark	\checkmark	\checkmark
Bans in restaurants x Mother's Educ	~	\checkmark	\checkmark	V
Observations	✓ 165,260	√ 165,260	v 161,697	✓ 165,260
Panel C: Robustness to beer tax				
Cig. Tax	1.06^{**}	1.07^{***}	0.44	-0.56*
Tax x (mom HS)	(0.46) -1.04* (0.52)	(0.30) -1.11*** (0.20)	(0.34) -0.33 (0.27)	(0.33) 0.49^{*} (0.25)
Tax x (mom >HS)	(0.52) -1.14** (0.52)	(0.39) -1.17***	(0.37) -0.43 (0.20)	(0.25) 0.55^{**} (0.27)
	(0.52)	(0.39)	(0.39)	(0.27)
Beer Tax x Mother's Educ Observations	\checkmark 165,260	$\sqrt{165,260}$	$\sqrt{161,697}$	\checkmark 165,260
Panel D: Robustness to current cigarette tax				
Cig. Tax	1.63***	1.24***	0.88*	-0.67*
Tax x (mom HS)	(0.58) -1.61**	(0.43) -1.30***	(0.45) -0.76	(0.40) 0.60^*
Tax x (mom >HS)	(0.65) -1.74**	(0.47) -1.36***	(0.49) -0.89	$(0.31) \\ 0.66^*$
	(0.68)	(0.47)	(0.55)	(0.34)
Current cigarette tax x Mother's Educ Observations	\checkmark 165,260	✓ 165,260	\checkmark 161,697	\checkmark 165,260
Cohorts	89-98	89-98	89-98	89-98
State FE	\checkmark	\checkmark	\checkmark	\checkmark
Time FE State space Cohort Trands	v	v	\checkmark	\checkmark
Baseline Controls	v V	v V	v √	v J

Table 10: Robustness of tax effect on educational outcomes to health policies

The cigarette tax variable corresponds to the mean of the cigarette tax rate in the 9 months following the estimated date of conception. It is deflated to real mid-1998 USD using a monthly consumer price index and scaled such that coefficients correspond to the effect of 10-cent cigarette tax increases. All outcome variables are dummies, coded as 0-100 for better readability so that coefficients can be interpreted as percentage points. All specifications control for state fixed effects, time fixed effects as well as statespecific time trends. In addition, all specifications control for gender, race and mother's age at giving birth. Panel A shows the baseline results. The remaining panels control for additional state-level health policies prevailing in the state of birth during the in-utero period (Panel B and Panel C) and in the state of residence in the year of being surveyed (Panel D). All additional controls are interacted with dummies for mother's education group. Robust standard errors clustered on state-level are in parenthesis. * p < 0.10, ** p < 0.05, *** p < 0.01. 45

	(1)	(2)	(3)	(4)
VARIABLES	Grade 9	enrolled	Grade9 school	Cogn. Diff.
Panel A: Baseline Results				
Cig. Tax	1.10**	0.88***	0.65*	-0.58*
Tax x (mom HS)	(0.44) -0.98*	(0.32) - 0.89^{**}	(0.35) -0.46	(0.33) 0.51^{**}
Tax x (mom >HS)	(0.50) -1.18**	(0.35) - 0.96^{***}	(0.37) - 0.66^*	(0.25) 0.57^{**}
Observations	(0.49) 165 260	(0.35) 165 260	(0.38) 161 697	(0.27) 165-260
	105,200	105,200	101,057	105,200
Panel B: Robustness to Education Expenditures				
Cig. Tax	0.86***	0.67***	0.48**	-0.44**
Toy y (mom HS)	(0.30)	(0.22) 0.70***	(0.23) 0.42*	(0.20) 0.41**
	(0.34)	(0.25)	(0.25)	(0.16)
$Tax x \pmod{>HS}$	-0.92***	-0.73***	-0.49*	0.45**
	(0.34)	(0.25)	(0.26)	(0.17)
Education exp. x Mother's Educ	.(.(.(.(
Observations	165,260	165,260	161,697	165,260
Panel C: Baseline, sample with SFR data				
Cig. Tax	1.08**	0.85**	0.63*	-0.61*
	(0.45)	(0.32)	(0.35)	(0.33)
Tax x (mom HS)	-0.97*	-0.87**	-0.45	0.52**
Torrer (more > IIC)	(0.51)	(0.35)	(0.37)	(0.25)
1ax x (mom > HS)	(0.50)	(0.35)	(0.39)	(0.38^{++})
	(0.00)	(0.00)	(0.00)	(0)
School Financing Principles x Mother's Educ Observations	× 164,062	x 164,062	× 160,518	× 164,062
Panel D: Robustness to school finance reforms				
Cig. Tax	0.74*	0.76*	0.31	-0.40
018. 100	(0.42)	(0.38)	(0.27)	(0.31)
Tax x (mom HS)	-0.63	-0.80*	-0.14	0.26
	(0.50)	(0.43)	(0.30)	(0.24)
$Tax x \pmod{>HS}$	-0.73	-0.87**	-0.22	0.34
	(0.47)	(0.42)	(0.32)	(0.25)
School Financing Principles x Mother's Educ	\checkmark	\checkmark	\checkmark	\checkmark
Observations	$164,\!062$	$164,\!062$	160,518	$164,\!062$
Cohorts	89-98	89-98	89-98	89-98
State FE	\checkmark	\checkmark	\checkmark	\checkmark
Time FE	\checkmark	\checkmark	\checkmark	\checkmark
State spec. Cohort Trends	 ✓ 	\checkmark	\checkmark	\checkmark
Baseline Controls	\checkmark	\checkmark	\checkmark	\checkmark

Table 11: Robustness of tax effect on educational outcomes to changes in the educ. system

The cigarette tax variable corresponds to the mean of the cigarette tax rate in the 9 months following the estimated date of conception. It is deflated to real mid-1998 USD using a monthly consumer price index and scaled such that coefficients correspond to the effect of 10-cent cigarette tax increases. All outcome variables are dummies, coded as 0-100 for better readability so that coefficients can be interpreted as percentage points. All specifications control for state fixed effects, time fixed effects as well as statespecific time trends. In addition, all specifications control for gender, race and mother's age at giving birth. Robust standard errors clustered on state-level are in parenthesis. * p < 0.10, ** p < 0.05, *** p < 0.01.

Main Figures



Figure 1: The evolution of tobacco taxes by state (in real 1998-Dollars, corresponding to PRAMS sample)



Figure 2: Socioeconomic differences in prenatal tobacco exposure and infant health



Figure 3: Socioeconomic differences in education at age 12 and 16



Figure 4: The heterogeneous effect of cigarette taxes on Grade 9 completion by age 16



Figure 5: The development of cigarette taxes in real 1998 - USD

A Additional Tables

Variable	Mean	Std. Dev.	Min.	Max.	Ν
Baby is female	0.489	0.5	0	1	684007
Age < =17	0.036	0.186	0	1	683247
Age 18-19	0.069	0.253	0	1	683247
Age 20-24	0.241	0.428	0	1	683247
Age 25-29	0.282	0.45	0	1	683247
Age 30-34	0.238	0.426	0	1	683247
Age 35-39	0.11	0.313	0	1	683247
Age $40+$	0.024	0.152	0	1	683247
Kessner Index: Adequate PNC	0.698	0.459	0	1	684026
Kessner Index: Intermediate PNC	0.196	0.397	0	1	684026
Kessner Index: Inadequate PNC	0.049	0.215	0	1	684026
Kessner Index: Unknown PNC	0.057	0.232	0	1	684026
Preg. Intention: later	0.178	0.382	0	1	661743
Preg. Intention: sooner	0.301	0.459	0	1	661743
Preg. Intention: now	0.406	0.491	0	1	661743
Preg. Intention: never	0.1	0.3	0	1	661743
Preg. Intention: not sure	0.015	0.121	0	1	661743
Mother has been or is married	0.64	0.48	0	1	683196
) stressors	0.283	0.45	0	1	677646
1 stressor	0.236	0.424	0	1	677646
2-4 stressors	0.368	0.482	0	1	677646
5-9 stressors	0.11	0.313	0	1	677646
10-14 stressors	0.004	0.062	0	1	677646
Other Asian	0.025	0.157	0	1	664653
White	0.745	0.436	0	1	664653
Black	0.165	0.371	0	1	664653
American Indian	0.01	0.101	0	1	664653
Chinese	0.006	0.08	0	1	664653
Japanese	0.002	0.045	0	1	664653
Filipino	0.006	0.079	0	1	664653
Hawaiian	0.002	0.05	0	1	664653
Other Nonwhite	0.026	0.158	0	1	664653
Alaska Native	0.002	0.044	0	1	664653
M. 1D	0.000	0.005	0	1	CC 4 CE 9

Table A.1: Summary statistics on additional controls: PRAMS sample

Outcome Variable Mother in HH	(1) age 8	(2) age 12	(3) age 16	(4) age 17	(5) age 18	(6) age 19	(7) age 22	$(8) \\ age 24$	(9) age 26
<i>a</i> : m	0.00	0.15	0.14	0.00		o T o	0.04	0.00**	0.04
Cig. Tax	0.08	0.17	-0.14	-0.02	0.85^{**}	0.78	0.34	-0.80***	-0.04
	(0.06)	(0.23)	(0.20)	(0.22)	(0.34)	(2.54)	(0.36)	(0.34)	(0.81)
	× /		· · · ·		× /	× /	· · · ·	· · · ·	· /
Observations	$179,\!155$	184,876	$192,\!852$	192,313	196,341	$179,\!692$	149,283	$144,\!235$	$141,\!100$
Cohorts	97-06	93-02	89-98	88-97	87-96	86-95	83-92	81-90	79-88
State FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Time FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
State spec. cohort trends	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Assessment year FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Race and Gender	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark

Table A.2: Sample selection

The outcome variable in all columns is a dummy taking on the value 100 if the observed individual lives in a household with his/her mother (as compared to a group quarter or a household without mother). The cigarette tax coefficient is expressed in real 1998-USD and scaled such that coefficients represent tax changes of ten cents. All regressions are based on our standard specification, controlling for state fixed effects, time fixed effects as well as state-specific time trends. Maternal characteristics are not being controlled for. Standard errors are clustered on state level.* p < 0.10, ** p < 0.05, *** p < 0.01.

Table 1.5. Hobustness of tax effect	on on th	outcomes t	o menusi	on or auditi		ar 1a0165
	(1)	(2)	(3)	(4)	(5)	(6)
VARIABLES	Smoker	Birth Weight	LBW	Prematurity	SGA 10th pctl.	Baby Alive
Panel A: Baseline Results						
Cig. Tax	-0.39^{***}	1.96^{***}	-0.07^{***}	-0.05	-0.08^{*}	-0.00
Tax x (mom HS)	(0.03) 0.27^{***}	-0.80* (0.45)	(0.02) 0.02 (0.02)	0.04*	(0.04) 0.02 (0.02)	0.00
Tax x (mom >HS)	(0.00) 0.36^{***}	-2.18*** (0.47)	(0.02) 0.04^{**}	(0.02) 0.04^{*} (0.02)	0.09**	-0.00
Observations	(0.12) 615,794	(0.47) 624,387	(0.02) 624,387	(0.02) 620,828	(0.04) 621,519	(0.00) 561,563
Panel B: Robustness to WIC						
Cig. Tax	-0.40***	2.06^{***}	-0.07***	-0.05	-0.08**	-0.01
Tax x (mom HS)	(0.08) 0.27^{***}	(0.63) -0.82^*	(0.02) 0.02	(0.04) 0.04	(0.04) 0.02 (0.02)	(0.01) 0.00
Tax x (mom $>$ HS)	(0.06) 0.36^{***} (0.11)	(0.47) -2.22*** (0.48)	(0.02) 0.04^* (0.02)	(0.02) 0.04^{*} (0.02)	(0.03) 0.09^{**} (0.04)	(0.00) -0.00 (0.00)
WIC x Mother's Educ	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Observations	610,527	619,053	619,053	615,529	616,218	556,711
Panel C: Robustness to Kessner Index						
Cig. Tax	-0.39^{***}	1.95^{***}	-0.07^{***}	-0.05	-0.08^{*}	-0.00
Tax x (mom HS)	(0.08) 0.26^{***} (0.06)	-0.76^{*}	(0.02) 0.02 (0.02)	(0.03) 0.04^{*} (0.02)	(0.04) 0.02 (0.03)	(0.01) 0.00 (0.00)
Tax x (mom $>$ HS)	(0.00) 0.36^{***} (0.12)	(0.43) -2.18^{***} (0.47)	(0.02) 0.04^{**} (0.02)	(0.02) 0.04^{*} (0.02)	(0.03) 0.09^{**} (0.04)	-0.00 (0.00)
Kessner Index x Mother's Educ	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Observations	$615,\!794$	$624,\!387$	$624,\!387$	620,828	621,519	$561,\!563$
Panel D: Robustness to Information						
Cig. Tax	-0.31***	1.83**	-0.06***	-0.05	-0.07*	-0.00
Tax x (mom HS)	(0.07) 0.23^{***}	(0.67) -0.61	(0.02) 0.02	(0.04) 0.03	(0.04) 0.02 (0.02)	(0.01) 0.00 (0.00)
Tax x (mom >HS)	(0.06) 0.31^{**} (0.13)	(0.44) -1.87*** (0.49)	(0.01) 0.04^{**} (0.02)	(0.02) 0.03^{*} (0.02)	(0.03) 0.08^{**} (0.04)	(0.00) -0.00 (0.00)
Doctor Info x Mother's Educ	√ F90.951	√ E 41 E90	√ F 41 F 90	√ ≂20.000	√ F 40, 200	√ 470.007
Observations	538,351	541,538	541,538	539,899	540,308	479,227
State FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Time FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
State spec. Cohort Trends Baseline Controls	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark

Table A.3:	Robustness	of tax	effect c	on birth	outcomes	to in	clusion	of	additional	control	variables
------------	------------	--------	----------	----------	----------	-------	---------	----	------------	---------	-----------

The cigarette tax variable corresponds to the mean of the cigarette tax rate in the 9 months following the estimated date of conception. It is deflated to real mid-1998 USD using a monthly consumer price index and scaled such that coefficients correspond to the effect of 10-cent cigarette tax increases. Columns 3-6 look at different dummy outcome variables. Dummies are coded as 0-100 for better readability so that coefficients can be interpreted as percentage points. The baseline specification is described below Table 4. Additional controls in this table are: i) a dummy for whether the mother received supplementary nutrition during her pregnancy ("WIC"), ii) dummies for whether the mother's prenatal care was rated as adequate, intermediate, inadequate or "unknown" on the birth certificate, iii) A dummy for whether the woman recalls being informed by her doctor about the dangers of smoking. All controls are interacted with three levels of maternal education. Panel A repeats the baseline results form Table 4. Panels B-D show results for the baseline specification plus different control variables. Robust standard errors clustered on state-level are in parenthesis. * p < 0.10, ** p < 0.05, *** p < 0.01.

VARIABLES	(1) Grade 9	(2) Dropout	(3) Grade9 school	(4) Cogn. Diff.
Panel A: Baseline Results				
Cig. Tax	1.10**	0.88***	0.65^{*}	-0.58*
Tax x (mom HS)	(0.44) -0.98*	(0.32) -0.89**	(0.35) -0.46	(0.33) 0.51**
	(0.50)	(0.35)	(0.37)	(0.25)
Tax x (mom $>$ HS)	-1.18^{**} (0.49)	-0.96^{***} (0.35)	-0.66^{*} (0.38)	0.57^{**} (0.27)
Observations	165,260	165,260	161,697	165,260
Panel B: Robustness to Family Income and Size				
Cig. Tax	1.11**	0.90***	0.63*	-0.54*
	(0.43)	(0.29)	(0.33)	(0.31)
$Tax x \pmod{HS}$	-1.02**	-0.92***	-0.47	0.48*
Tax x (mom >HS)	(0.48) -1 25**	(0.32) _0 99***	(0.35) -0.68*	(0.24) 0.54**
$\max x \pmod{>nS}$	(0.47)	(0.32)	(0.37)	(0.25)
(Family Income, Size) x Mother's Educ	1	1	\checkmark	\checkmark
Observations	165,260	165,260	161,697	165,260
Panel C: Robustness to Family's rel. Income				
Cig. Tax	1.12**	0.89***	0.66**	-0.61*
T (H ())	(0.44)	(0.32)	(0.33)	(0.32)
Tax x (mom HS)	-1.02^{**} (0.50)	-0.90** (0.35)	-0.48	(0.53^{**})
Tax x (mom >HS)	-1.22**	-0.97***	-0.68*	0.61**
Family's rol Income y Mother's Educ	(0.49)	(0.35)	(0.37)	(0.26)
Observations	v 165,260	v 165,260	v 161,697	165,260
Panel D: Robustness mother's employment				
Cig. Tax	1.04**	0.84***	0.62*	-0.54
	(0.44)	(0.31)	(0.34)	(0.33)
$Tax x \pmod{HS}$	-0.92^{*}	-0.86**	-0.43	0.46^{*}
$T_{ax} \times (mom > HS)$	(0.50) 1 19**	(0.33)	(0.37)	(0.24) 0.52**
	(0.49)	(0.34)	(0.39)	(0.26)
Mother's Employment x Mother's Educ	\checkmark	\checkmark	\checkmark	\checkmark
Observations	$165,\!260$	$165,\!260$	$161,\!697$	165,260
Panel E: Robustness father's employment				
Cig. Tax	1.02	1.10*	0.26	-0.76***
Tax x (mom HS)	(0.62)	(0.58)	(0.38)	(0.26) 0.67**
1a. A (110111 11.5)	(0.70)	(0.62)	(0.39)	(0.25)
$Tax x \pmod{>HS}$	-1.15*	-1.24*	-0.33	0.72**
	(0.67)	(0.63)	(0.39)	(0.28)
Father's Employment x Mother's Educ	\checkmark	\checkmark	\checkmark	\checkmark
Observations	$122,\!024$	$122,\!024$	119,731	122,024
State FE	\checkmark	\checkmark	\checkmark	\checkmark
Time FE	\checkmark	\checkmark	\checkmark	\checkmark
State spec. Cohort Trends	~	√	\checkmark	\checkmark
Baseline Controls	\checkmark	\checkmark	1	1

Table A.4: Robustness of tax effect on educational outcomes to family's economic situation

The cigarette tax variable corresponds to the mean of the cigarette tax rate in the 9 months following the estimated date of conception. It is deflated to real mid-1998 USD using a monthly consumer price index and scaled such that coefficients correspond to the effect of 10-cent cigarette tax increases. All outcome variables are dummies, coded as 0-100 for better readability so that coefficients can be interpreted as percentage points. All specifications control for state fixed effects, time fixed effects as well as state-specific time trends. In addition, all specifications control for gender, race, mother's education and her age at giving birth. Robust standard errors clustered on state-level are in parenthesis. * p < 0.10, ** p < 0.05, *** p < 0.01.

	(1)	(2)	(3)	(4)
VARIABLES	Grade 9	enrolled	Grade9 school	Cogn. Diff.
Panel A: Baseline Results				
Cig. Tax	1.10**	0.88***	0.65^{*}	-0.58*
Tax x (mom HS)	(0.44) - 0.98^*	(0.32) - 0.89^{**}	(0.35) - 0.46	(0.33) 0.51^{**}
Tax x (mom $>$ HS)	(0.50) -1.18**	(0.35) - 0.96^{***}	(0.37) - 0.66^*	(0.25) 0.57^{**}
Observations	(0.49) 165,260	(0.35) 165,260	(0.38) 161,697	(0.27) 165,260
Panel B: Robustness to Medicaid eligibility				
Cig. Tax	0.83**	0.72**	0.46	-0.54*
Tax x (mom HS)	(0.39) -0.71 (0.42)	(0.31) -0.76**	(0.29) -0.27 (0.20)	(0.30) 0.45^{*} (0.25)
Tax x (mom $>$ HS)	(0.43) -0.87** (0.40)	(0.33) -0.78** (0.32)	(0.29) -0.44 (0.30)	(0.25) 0.53^{**} (0.25)
Medicaid elig. thresh. x Mother's Educ Observations	\checkmark 165,260	\checkmark 165,260	✓ 161,697	$\checkmark 165,260$
Panel C: Robustness to welfare reform				
Cig. Tax	0.94**	0.83**	0.50	-0.56*
Tax x (mom HS)	(0.42) -0.83* (0.48)	(0.33) -0.84** (0.35)	(0.31) -0.32 (0.33)	(0.32) 0.48^{*} (0.25)
Tax x (mom >HS)	(0.46) -0.96^{**} (0.45)	(0.35) -0.89^{**} (0.35)	(0.33) -0.45 (0.33)	(0.25) 0.53^{**} (0.25)
Welfare Reform x Mother's Educ Observations	$\checkmark 165,260$	✓ 165,260	✓ 161,697	$\checkmark 165,260$
Panel D: Robustness to the business cycle				
Cig. Tax	0.99^{**}	0.70^{**}	0.66^{*}	-0.54^{*}
Tax x (mom HS)	(0.43) -0.85^{*} (0.46)	(0.23) -0.71^{**} (0.30)	(0.36) -0.46 (0.36)	(0.30) 0.46^{*} (0.24)
Tax x (mom $>$ HS)	(0.10) -1.07** (0.45)	(0.30) -0.78^{**} (0.30)	(0.30) -0.67^{*} (0.37)	(0.21) (0.52^{**}) (0.25)
Unemployment rate x Mother's Educ	\checkmark	\checkmark	\checkmark	\checkmark
Observations Cohorts	165,260	165,260	161,697	165,260
State FE	09-90 V	09-90 V	09-90 V	09-90 V
Time FE	· √	↓	• •	↓ √
State spec. Cohort Trends	√	√	√	√
Race and Gender	\checkmark	\checkmark	\checkmark	\checkmark
Mother's age at giving birth	\checkmark	\checkmark	\checkmark	\checkmark

Table A.5: Robustness of tax effect on educational outcomes to state-level policy changes and to the business cycle

The cigarette tax variable corresponds to the mean of the cigarette tax rate in the 9 months following the estimated date of conception. It is deflated to real mid-1998 USD using a monthly consumer price index and scaled such that coefficients correspond to the effect of 10-cent cigarette tax increases. All outcome variables are dummies, coded as 0-100 for better readability so that coefficients can be interpreted as percentage points. All specifications control for state fixed effects, time fixed effects as well as statespecific time trends. In addition, all specifications control for gender, race and mother's age at giving birth. Robust standard errors clustered on state-level are in parenthesis. * p < 0.10, ** p < 0.05, *** p < 0.01.

	(1)	(2)	(3)	(4)
VARIABLES	Grade 9	enrolled	Grade9 school	Cogn. Diff.
Panel A: Baseline Results				
Cig. Tax	1.10**	0.88***	0.65*	-0.58*
Tax x (mom HS)	(0.44) - 0.98^*	(0.32) - 0.89^{**}	$(0.35) \\ -0.46$	(0.33) 0.51^{**}
$Tax \times (mom > HS)$	(0.50) -1.18**	(0.35) - 0.96^{***}	(0.37) - 0.66^*	(0.25) 0.57^{**}
Observations	(0.49)	(0.35)	(0.38)	(0.27)
	105,200	105,200	101,097	105,200
Panel B: Robustness to Food Stamps				
Cig. Tax	1.17**	0.88**	0.73**	-0.63*
Tax x (mom HS)	(0.45)	(0.33) -0.92**	(0.31)	(0.33) 0.53**
	(0.51)	(0.36)	(0.34)	(0.24)
$Tax \ge (mom > HS)$	-1.20**	-0.97***	-0.67*	0.60**
	(0.49)	(0.36)	(0.34)	(0.25)
Food stamp exp. x Mother's Educ	1	1		1
Observations	165,260	165,260	161,697	165,260
Panel C: Robustness to Income Maintenance				
Cig. Tax	1.01**	0.68**	0.68*	-0.44
	(0.43)	(0.33)	(0.36)	(0.35)
Tax x (mom HS)	-0.76^{+}	-0.67^{*}	-0.38	(0.42)
$Tax x \pmod{>HS}$	-0.96**	-0.72**	-0.59	(0.20) 0.48*
	(0.44)	(0.34)	(0.36)	(0.26)
			. ,	
Income maintenance exp. x Mother's Educ	√ 165 960	√ 165.960	161 607	√ 165.960
Observations	105,200	105,200	161,097	105,200
Panel D: Robustness to Pub. Medical Ass.				
Cig. Tax	1.43***	1.06***	0.86**	-0.74*
	(0.49)	(0.36)	(0.35)	(0.38)
$Tax x \pmod{HS}$	-1.31**	-1.09***	-0.66*	0.64**
Tour a (more > HC)	(0.56)	(0.40)	(0.39)	(0.30)
1ax x (mom > 115)	(0.55)	(0.39)	(0.41)	(0.73)
	· · /		× /	
Pub. med. ass. exp. x Mother's Educ	107.000	107.000	161 607	107.000
Observations	105,200	165,260	161,097	105,200
Cohorts	89-98	89-98	89-98	89-98
State FE	√	√	\checkmark	✓
Time FE State grade Cohort Trands	\checkmark	\checkmark	\checkmark	\checkmark
Bace and Gender	v ./	v ./	V	∨ ./
Mother's age at giving birth	• •	√ √	√	↓

Table A.6: Robustness of tax effect on educational outcomes to state transfers

The cigarette tax variable corresponds to the mean of the cigarette tax rate in the 9 months following the estimated date of conception. It is deflated to real mid-1998 USD using a monthly consumer price index and scaled such that coefficients correspond to the effect of 10-cent cigarette tax increases. All outcome variables are dummies, coded as 0-100 for better readability so that coefficients can be interpreted as percentage points. All specifications control for state fixed effects, time fixed effects as well as statespecific time trends. In addition, all specifications control for gender, race and mother's age at giving birth. Robust standard errors clustered on state-level are in parenthesis. * p < 0.10, ** p < 0.05, *** p < 0.01.

Β Calculation of duration of pregnancy in months

In the PRAMS data set, length of gestation is reported in five categories each of which is spanning several weeks. In order to assign a cigarette tax rate to each pregnancy, as described in section 3.3.1, we map a gestational length measured in months to each of the five categories. This is illustrated in Table B.1 which at the same time gives an overview of the distribution of pregnancies over the five categories. The gestational age categories reported on birth certificates are calculated starting from the beginning of the menstrual cycle during which the woman got pregnant. They therefore include the approximately two weeks before the pregnancy started. When converting gestational age categories into months we take this into account. When information on the length of gestation was missing, we assumed a standard length of nine months.

Table B.1: Mapping of gestational age categories to monthly duration of pregnancy						
Gestational age category	Share of pregnancies in $\%$	sd	assigned duration in months			
<27 weeks 28-33 weeks	0.466 1.626	$6.808 \\ 12.65$	6 months 7 months			
34-36 weeks 37-42 weeks	5.866 91.77	23.50 27.48	8 months 9 months			
43+ weeks gestational age missing	$0.268 \\ 0.821$	$5.173 \\ 9.024$	10 months 9 months			

\mathbf{C} Calculation of price elasticities

Price elasticities of smoking participation based on Evans and Ringel (1999) are calculated as follows: $e_p = \frac{\frac{\partial S}{\partial T}}{\frac{\partial P}{\partial T}} \frac{\bar{P}}{S}$. In this equation, $\frac{\partial S}{\partial T}$ is the point estimate of the tax impact on smoking participation, i.e. the estimated tax coefficient. $\frac{\partial P}{\partial T}$ is the pass-through rate of cigarette taxes on retail prices. Based on the existing literature we use the value 1.15 (Evans and Ringel, 1999) which is supported by a large number of other sources as well (Sumner, 1981; Sullivan, 1985; Keeler et al., 1996; Federal Trade Commission, 1997; Evans et al., 1999; Gruber and Köszegi, 2001). \bar{P} is the average price of cigarettes. We use data provided by Orzechowski and Walker (2014) on yearly average cigarette prices per state and adjust these to the same base period as our tax data (1998). We then calculate the average price experienced by each mother during her pregnancy and average across all individuals in our sample. \bar{S} is the average rate of smokers in the sampling period. The calculation of price elasticities are displayed in Table C.1.

Table C.1. Calculation of price elasticities of smoking participation					
	Full sample	$\mathrm{mom} < \mathrm{HS}$	$\operatorname{mom}\operatorname{HS}$	$\mathrm{mom} > \mathrm{HS}$	
Estimated coefficient $\frac{\partial S}{\partial T}$ Pass-through rate $\frac{\partial P}{\partial T}$ Average price \bar{P} Average smoking \bar{S}	-0.12 1.15 4.67 11.62	-0.39 1.15 4.67 20.91	-0.12 1.15 4.67 16.80	-0.03 1.15 4.67 5.45	
Resulting price elasticity	-0.42	-0.76	-0.29	-0.23	

Table C 1: Calculation of price electicities of smalring participation

Calculation of labor market effects from cigarette tax D increases

This section illustrates the calculation of the estimated long-term labor market effect of cigarette taxes which are mentioned in Section 7. Our regression results suggest that a 10 cent tax increase leads to a 1.1 percentage points higher chance of having completed grade 9 by the age of 16 for somebody from the lowest socioeconomic group. A student who has completed grade 9 by the age of 16 has a roughly 15 percent higher chance of completing high school instead of dropping out, based on representative statistics from the U.S. (Barro and Kolstad, 1987).⁵⁸ Based on these numbers, we can calculate the intention-to-treat effect of a 10 cent tax increase on high school completion:

ITT(High School) = 0.011 * 0.15 = 0.00165

A 10 cent tax increase leads to a 0.17 percentage point increase in the probability of completing high school of the exposed cohorts from the lowest socioeconomic group. How does this translate into later life outcomes? In terms of expected income, a high school degree compared to no high school degree has been estimated to lead to gains in median annual earnings of 10,000 USD in the year 2013 for men and 8,500 USD for women (Snyder et al., 2016). In addition, using labor market statistics from 2013, the chance of being employed for somebody with a high school degree as compared to a high school dropout was 20 percentage points higher (51 percent vs. 31 percent) (Snyder et al., 2016). Both components lead to expected increases in annual earnings associated with a high school degree as follows:

 $^{^{58}}$ This number is based on statistics from the High School and Beyond - Survey (HS&B) sophomore cohort of 1980. The HS&B data set is representative for pupils enrolled in public schools in the United States. According to these numbers, students who were around 14 at the moment of starting grade 9 had a roughly 15 percent smaller chance of dropping out of high school than students who were one year older when starting grade 9 (24 percent vs. 9 percent according to Barro and Kolstad (1987), p.53). An alternative measure based on the same data is the difference in dropping out from high school of those that have ever been retained (27.2 percent) compared to those that have never been retained (12.4 percent). The resulting difference is around 15 percentage points as well.

 $Income gain = \begin{cases} 0.2 * median \ earning \ men + HS \ mark-up = 0.2 * 40,000 + 10,000 = 18,000 \ for \ men \\ 0.2 * median \ earning \ women + HS \ mark-up = 0.2 * 31,000 + 8,500 = 14,700 \ for \ women \end{cases}$

The corresponding ITT effect from a 10 cent cigarette tax in terms of additional earnings in the lowest socioeconomic group is as follows:

ITT (Earnings) =
$$\begin{cases} 0.00165 * 18,000 = 29.70 \text{ for men} \\ 0.00165 * 14,700 = 24.26 \text{ for women.} \end{cases}$$

IAAEU Discussion Paper Series in Economics

01/2012	Relative Consumption Concerns or Non-Monotonic Preferences?
	Inga Hillesheim and Mario Mechtel
02/2012	Profit Sharing and Relative Consumption
	Laszlo Goerke
	[published as: Goerke, L. (2013). Profit Sharing and Relative Consumption. <i>Economics Letters</i> 118, 167-169.]
03/2012	Conspicuous Consumption and Communism: Evidence From East and West Germany
	Tim Friehe and Mario Mechtel
	[published as: Friehe, T. and M. Mechtel (2014). Conspicuous Consumption and Political Regimes: Evidence from East and West Germany. <i>European Economic Review</i> 67, 62-81.]
04/2012	Unemployment Benefits as Redistribution Scheme for Trade Gains - A Positive Analysis
	Marco de Pinto
	[published as: de Pinto, M. (2013). International Trade and Unemployment: on the Redistribution of Trade Gains when Firms Matter, Physica-Verlag (Springer), Berlin.]
05/2012	Failure of Ad Valorem and Specific Tax: Equivalence under Uncertainty
	Laszlo Goerke, Frederik Herzberg and Thorsten Upmann
	[revised version published as: Goerke, L., F. Herzberg and T. Upmann (2014). Failure of Ad Valorem and Specific Tax Equivalence under Uncertainty. <i>International Journal of Economic</i> <i>Theory</i> 10, 387-402.]
06/2012	The Redistribution of Trade Gains and the Equity-Efficiency Trade-Off
	Marco de Pinto
	[published as: de Pinto, M. (2013). International Trade and Unemployment: on the Redistribution of Trade Gains when Firms Matter, Physica-Verlag (Springer), Berlin.]
07/2012	Trade Union Membership and Sickness Absence: Evidence from a Sick Pay Reform
	Laszlo Goerke and Markus Pannenberg
	[published as: Goerke, L. and M. Pannenberg (2015). Trade Union Membership and Sickness
	Absence: Evidence from a Sick Pay Reform. Labour Economics 33, 13-25.]
08/2012	Risk-Sorting and Preference for Team Piece Rates
	Agnes Bäker and Vanessa Mertins
	[published as: Bäker, A. and V. Mertins (2013). Risk-Sorting and Preference for Team Piece Rates. <i>Journal of Economic Psychology</i> 34, 285-300.]
09/2012	Union Wage Setting and International Trade
	Hartmut Egger and Daniel Etzel
	[published as: Egger, H. and D. Etzel (2014). Union wage-setting and international trade with footloose capital. <i>Regional Science and Urban Economics</i> 48, 56-67.]

10/2012	How Much Do Others Matter? Explaining Positional Concerns for Different Goods and
	Inga Hillesheim and Mario Mechtel [published as: Hillesheim, I. and M. Mechtel (2013). How Much Do Others Matter? Explaining Positional Concerns for Different Goods and Personal Characteristics. <i>Journal of Economic</i> <i>Psychology</i> 34, 61-77.]
11/2012	Benefit Morale and Cross-Country Diversity in Sick Pay Entitlements
	<i>Daniel Arnold</i> [published as: Arnold, D. (2013). Benefit Morale and Cross-Country Diversity in Sick Pay Entitlements. <i>Kyklos</i> 66, 27-45.]
01/2013	Relative Consumption and Tax Evasion
	Laszlo Goerke [published as: Goerke, L. (2013). Relative Consumption and Tax Evasion. Journal of Economic Behavior & Organization 87, 52-65.]
02/2013	Variants of the Monoamine Oxidase A Gene (MAOA) Predict Free-riding Behavior in Women in a Strategic Public Goods Experiment
	Vanessa Mertins, Andrea B. Schote and Jobst Meyer [published as: Mertins, V., A.B. Schote and J. Meyer (2013). Variants of the Monoamine Oxidase A Gene (MAOA) Predict Free-riding Behavior in Women in a Strategic Public Goods Experiment. Journal of Neuroscience, Psychology, and Economics 6, 97-114.]
03/2013	Direct Evidence on Income Comparisons and Subjective Well-Being Laszlo Goerke and Markus Pannenberg
04/2013	Flexibilisation without Hesitation? Temporary Contracts and Workers' Satisfaction Adrian Chadi and Clemens Hetschko [forthcoming as: Chadi, A and C. Hetschko. Flexibilisation without Hesitation? Temporary Contracts and Job Satisfaction. Oxford Economic Papers.]
05/2013	Structural and Cyclical Effects of Tax Progression Jana Kremer and Nikolai Stähler
06/2013	Regional Unemployment and Norm-Induced Effects on Life Satisfaction Adrian Chadi [published as: Chadi, A. (2014). Regional Unemployment and Norm-Induced Effects on Life Satisfaction. <i>Empirical Economics</i> 46, 1111-1141.]
07/2013	Third Person Effects in Interview Responses on Life Satisfaction <i>Adrian Chadi</i> [published as: Chadi, A. (2013). Third Person Effects in Interview Responses on Life Satisfaction. <i>Journal of Applied Social Science Studies (Schmollers Jahrbuch</i>) 133, 323-333.]
08/2013	The Role of Task Meaning on Output in Groups: Experimental Evidence Agnes Bäker and Mario Mechtel
09/2013	Gender Differences in Responsiveness to a Homo Economicus Prime in the Gift-Exchange Game <i>Vanessa Mertins and Susanne Warning</i>

10/2013	Relative Consumption, Working Time, and Trade Unions
	Laszlo Goerke and Inga Hillesheim [published as: Goerke, L. and I. Hillesheim (2013). Relative Consumption, Working Time, and Trade Unions. Labour Economics 24, 170-179.]
11/2013	The Role of Interviewer Encounters in Panel Responses on Life Satisfaction
	Adrian Chadi [published as: Chadi, A. (2013). The Role of Interviewer Encounters in Panel Responses on Life Satisfaction. <i>Economics Letters</i> 121, 550-554.]
12/2013	It's the Occupation, Stupid! Explaining Candidates' Success in Low-Information Elections
	<i>Mario Mechtel</i> [published as: Mechtel, M. (2014). It's the occupation, stupid! Explaining candidates' success in low-information elections. <i>European Journal of Political Economy</i> 33, 53-70.]
13/2013	Do Overconfident Workers Cooperate Less? The Relationship between Overconfidence and Cooperation in Team Production
	Vanessa Mertins and Wolfgang Hoffeld [published as: Mertins, V. and W. Hoffeld (2015). Do Overconfident Workers Cooperate Less? The Relationship between Overconfidence and Cooperation in Team Production. <i>Managerial</i> <i>and Decision Economics</i> 36, 265-274.]
01/2014	Income Tax Buyouts and Income Tax Evasion
	Laszlo Goerke [published as: Goerke, L. (2015). Income Tax Buyouts and Income Tax Evasion. International Tax and Public Finance 22, 120-143.]
02/2014	Family Employees and Absenteeism <i>Jörn Block, Laszlo Goerke, José María Millán and Concepción Román</i> [published as: Block, J., L. Goerke, J.M. Millán and C. Román (2014). Family employees and absenteeism. <i>Economics Letters</i> 123, 94-99.]
03/2014	Dissatisfied with Life or with Being Interviewed? Happiness and Motivation to Participate in a Survey Adrian Chadi
04/2014	Gambling to Leapfrog in Status? <i>Tim Friehe and Mario Mechtel</i> [forthcoming as Friehe, T. and M. Mechtel. Gambling to Leapfrog in Status. <i>Review of</i> <i>Economics of the Household</i> .]
05/2014	The Magic of the New: How Job Changes Affect Job Satisfaction Adrian Chadi and Clemens Hetschko
06/2014	The Labor Market Effects of Trade Unions – Layard Meets Melitz <i>Marco de Pinto and Jochen Michaelis</i> [published as: de Pinto, M. and J. Michaelis. The Labor Market Effects of Trade Unions – Layard Meets Melitz. <i>International Economics and Economic Policy</i> 13(2), 223-232.]
07/2014	Workers' Participation in Wage Setting and Opportunistic Behavior: Evidence from a Gift-Exchange Experiment Jörg Franke, Ruslan Gurtoviy and Vanessa Mertins

08/2014	Wage Delegation in the Field
	Sabrina Jeworrek and Vanessa Mertins
09/2014	Tax Evasion by Individuals
	Laszlo Goerke
	[published as: Goerke, L. Tax Evasion by Individuals. <i>Encyclopedia of Law and Economics:</i> Springer Reference.]
10/2014	Sickness Absence and Works Councils
	Daniel Arnold, Tobias Brändle and Laszlo Goerke
	[forthcoming as: Arnold, D., T. Brändle and L. Goerke. Sickness Absence and Works Councils – Evidence from German Individual and Linked Employer-Employee Data. <i>Industrial Relations: A</i>
	Journal of Economy and Society.]
11/2014	Positional Income Concerns: Prevalence and Relationship with Personality and Economic Preferences
	Tim Friehe, Mario Mechtel and Markus Pannenberg
12/2014	Unionization, Information Asymmetry and the De-location of Firms
	Marco de Pinto and Jörg Lingens
01/2015	The One Constant: A Causal Effect of Collective Bargaining on Employment Growth?
01/2013	Evidence from German Linked-Employer-Employee Data
	Tobias Brändle and Laszlo Goerke
02/2015	How Job Changes Affect People's Lives – Evidence from Subjective Well-being Data
·	Adrian Chadi and Clemens Hetschko
03/2015	Concerns about the Euro and Happiness in Germany during Times of Crisis
	Adrian Chadi
	[published as: Chadi, A. Concerns about the Euro and Happiness in Germany during Times of Crisis. <i>European Journal of Political Economy</i> 40, 126-146.]
04/2015	Missing at Work – Sickness-related Absence and Subsequent Job Mobility
	Adrian Chadi and Laszlo Goerke
05/2015	Social Identity and Social Free-Riding
	Mark Bernard, Florian Hett and Mario Mechtel
	[published as: Bernard, M., F. Hett and M. Mechtel. Social Identity and Social Free-Riding. European Economic Review 90, 4-17.]
06/2015	Peer Settings Induce Cheating on Task Performance
	Agnes Bäker and Mario Mechtel
07/2015	The Protestant Fiscal Ethic: Religious Confession and Euro Skepticism in Germany
-	Adrian Chadi and Matthias Krapf
	[forthcoming as: Chadi, A and Krapf, M. The Protestant Fiscal Ethic: Religious Confession and Euro Skepticism in Germany. <i>Economic Inquiry</i> .]

08/2015	Firm-level versus Sector-level Trade Unions – The Role of Rent-Sharing Motives <i>Marco de Pinto</i> [forthcoming as: de Pinto, M. The Impact of Unionization Structures with Heterogeneous Firms and Rent-Sharing Motives. <i>Scandinavian Journal of Economics</i> .]
09/2015	Direct Evidence for Income Comparisons and Subjective Well-Being across Reference Groups
	Laszlo Goerke and Markus Pannenberg
	[published as: Goerke, L. and M. Pannenberg. Direct Evidence for Income Comparisons and Subjective Well-Being across Reference Groups. <i>Economics Letters</i> 137, 95-101.]
10/2015	Leadership and persistency in spontaneous dishonesty
	Susanne Braun and Lars Hornuf
11/2015	How are Work-related Characteristics Linked to Sickness Absence and Presenteeism? – Theory and Data –
	Daniel Arnold and Marco de Pinto [published as: Arnold, D. and M. de Pinto. How are Work-related Characteristics Linked to Sickness Absence and Presenteeism? – Theory and Data. <i>Journal of Applied Social Science</i> <i>Studies (Schmollers Jahrbuch)</i> 135(4), 465-498.]
01/2016	Paid Vacation Use: The Role of Works Councils
,	Laszlo Goerke and Sabrina Jeworrek
02/2016	Identification of Attrition Bias Using Different Types of Panel Refreshments Adrian Chadi
03/2016	Welfare-enhancing Trade Unions in an Oligopoly with Excessive Entry
	Marco de Pinto and Laszlo Goerke
04/2016	Sick Pay Reforms and Health Status in a Unionised Labour Market
	[forthcoming as: Goerke, L. Sick Pay Reforms and Health Status in a Unionised Labour Market. Scottish Journal of Political Economy.]
05/2016	Tax Evasion in a Cournot Oligopoly with Endogenous Entry
	Laszlo Goerke [forthcoming as: Goerke, L. Tax Evasion in a Cournot Oligopoly with Endogenous Entry. International Tax and Public Finance.]
06/2016	The Emergence of the Global Fintech Market: Economic and Technological Determinants Christian Haddad and Lars Hornuf
01/2017	The Impact of Unionization Costs when Firm-selection Matters Marco de Pinto and Jörg Lingens
02/2017	Can Television Reduce Xenophobia? The Case of East Germany Lars Hornuf and Marc Oliver Rieger

03/2017 The effect of cigarette taxes during pregnancy on educational outcomes of the next generation Sonja Settele and Reyn van Ewijk